Rochester Institute of Technology

RIT Digital Institutional Repository

Theses

8-1-2013

Stability analysis of switched dc-dc boost converters for integrated circuits

Kevin Fronczak

Follow this and additional works at: https://repository.rit.edu/theses

Recommended Citation

Fronczak, Kevin, "Stability analysis of switched dc-dc boost converters for integrated circuits" (2013). Thesis. Rochester Institute of Technology. Accessed from

This Thesis is brought to you for free and open access by the RIT Libraries. For more information, please contact repository@rit.edu.

$R^{\cdot}I \cdot T$

Stability Analysis of Switched DC-DC Boost Converters for Integrated Circuits

by

Kevin C. Fronczak

A Thesis Submitted in Partial Fulfillment of the Requirements for the Degree of MASTER OF SCIENCE

in Electrical Engineering

DEPARTMENT OF ELECTRICAL AND MICROELECTRONIC ENGINEERING KATE GLEASON COLLEGE OF ENGINEERING ROCHESTER INSTITUTE OF TECHNOLOGY ROCHESTER, NEW YORK AUGUST, 2013

Approved by:

Dr. Robert J. Bowman, Thesis Adviser

Dr. James E. Moon, Committee Member

Dr. Karl Hirschman, Committee Member

Dr. Sohail A. Dianat, Department Head

Date

Date

Date

Date

Stability Analysis of Switched DC-DC Boost Converters for Integrated Circuits

Kevin C. Fronczak Thesis Adviser: Dr. Robert J. Bowman

Abstract

Boost converters are very important circuits for modern devices, especially batteryoperated integrated circuits. This type of converter allows for small voltages, such as those provided by a battery, to be converted into larger voltage more suitable for driving integrated circuits.

Two regions of operation are explored known as Continuous Conduction Mode and Discontinuous Conduction Mode. Each region is analyzed in terms of DC and small-signal performance. Control issues with each are compared and various error amplifier architectures explored. A method to optimize these amplifier architectures is also explored by means of Genetic Algorithms and Particle Swarm Optimization.

Finally, stability measurement techniques for boost converters are explored and compared in order to gauge the viability of each method. The Middlebrook Method for measuring stability and cross-correlation are explored here.

ACKNOWLEDGMENTS

This project would not be possible without the support of many individuals. I would like to thank everyone who has inspired me and guided me throughout my studies at RIT.

I would like to thank my adviser, Dr. Robert Bowman, for his very helpful guidance that has helped me to grow academically. His passion for analog design is unrivaled, and that enthusiasm is evident in the fantastic courses he has taught.

I would like to express my gratitude to Synaptics Inc. for the research funds to allow me to complete this work, as well as providing me with the opportunity to work in such an exciting field. In particular, I'd like to thank Mark Pude and Imre Knauz whose guidance and encouragement enabled me to produce the best work possible. Thank you to Murat Ozbas, whose encouragement to properly budget time between academics and work has allowed me to complete this thesis.

To my parents Greg and Barb, I thank you for providing endless love and support - I would have never gotten to this point were it not for you. Finally, I thank my beautiful fiancée Allegra whose love and support could only be outmatched by her patience as she endured this long process with me.

Contents

1	Intr	oduction	1
	1.1	Importance of Stability	1
	1.2	Thesis Organization	2
2	Boo	st Converter DC Characteristics	4
	2.1	Continuous Conduction Mode (CCM)	5
		2.1.1 Ideal Converter	5
		2.1.2 Non-Ideal Converter	7
	2.2	Discontinuous Conduction Mode (DCM) 1	3
		2.2.1 Ideal Converter	3
		2.2.2 Non-Ideal Converter	5
	2.3	CCM vs. DCM	8
3	\mathbf{Sm}	ll-Signal Analysis 2	4
	3.1	PWM Generation	5
	3.2	State-Space Averaging 2	7
	3.3	Average PWM Model 3	1
		3.3.1 PWM Switch in CCM	2
		3.3.2 PWM Switch in DCM	4
	3.4	Transfer Functions	9
		3.4.1 CCM: Ideal Transfer Function	0
		3.4.2 CCM: Non-Ideal Transfer Function	2
		3.4.3 DCM: Ideal Transfer Function	6
		3.4.4 DCM: Non-Ideal Transfer Function	8
	3.5	Transfer Function Comparison	3

4	Cor	nverter Control	6
	4.1	Error Amplifiers	(
		4.1.1 Op-Amps vs. OTAs	
		4.1.2 Lag Compensation (Proportional-Integral)	
		4.1.3 Lag Compensation Plus Pole	
		4.1.4 Lag Compensation Plus Pole with Feedback	
		4.1.5 Lead Compensation (Proportional-Derivative)	
		4.1.6 Lag-Lead Compensation	
		$(Proportional-Integral-Derivative) \ . \ . \ . \ . \ . \ . \ . \ . \ . \ $	
	4.2	Controller Selection	
5	Full	Boost Converter Model Analysis	
	5.1	CCM Converter Model	
		5.1.1 Component Sensitivity	
	5.2	DCM Converter Model	
		5.2.1 Component Sensitivity	
	5.3	Performance Comparisons	
6	Sta	bility Measurement Techniques	
	6.1	Step Response	
	6.2	Frequency Response	
		6.2.1 Injecting a Voltage	
		6.2.2 Injecting a Current	
		6.2.3 Response Measurement	
	6.3	Frequency Response via Cross-Correlation	1
		6.3.1 Cross-Correlation $\ldots \ldots \ldots$	1
		6.3.2 Pseudo-Random Binary Sequences (PRBS)	1
		6.3.3 Implementation of a PRBS Cross-Correlation Method	1
	6.4	Stability Measurement Comparisons	1
7	Cor	nclusion	1
	7.1	Future Work	1
Λ,	nnor	dicos	1,
د ب	hhcn		-

\mathbf{A}	Use of Optimization Algorithms in Controller Design			109
	A.1	Geneti	c Algorithms	110
		A.1.1	Queen-Bee Genetic Algorithm (QBGA)	111
		A.1.2	Fitness Function	113
		A.1.3	QBGA Results	115
	A.2	Particl	e Swarm Optimization	115
		A.2.1	PSO Results	117
	A.3	QBGA	and PSO Performance Comparison	117
В	Pow	ver Sup	oply Simulation Tutorial with MATLAB [®]	119

List of Figures

2.1	Ideal Boost Converter Circuit	4
2.2	Inductor Current in Boost Converter for CCM and DCM \hdots	5
2.3	Non-Ideal Boost Converter Circuit	8
2.4	Normalized Power Loss in CCM and DCM	9
2.5	Output Voltage versus Duty Cycle Comparison for DCM and CCM $$.	19
2.6	Efficiency versus Output Current Comparison for DCM and CCM $$	20
2.7	Output Ripple Voltage versus Duty Cycle Comparison for DCM and	
	CCM	20
2.8	Sensitivity of Efficiency to Changing r_L for CCM and DCM \ldots	21
2.9	Sensitivity of Efficiency to Changing C_{SW} for CCM and DCM \ldots	21
3.1	Regulated DC Converter Block Diagram (Adapted from $[13]$)	24
3.2	PWM Generation Circuit	25
3.3	PWM Generation Circuit Waveforms	26
3.4	Ideal Boost Converter with Switch Closed	28
3.5	Ideal Boost Converter with Switch Open	28
3.6	PWM Switch (adapted from $[18]$ and $[19]$)	32
3.7	Ideal Boost Converter with Common-Active PWM Switch	33
3.8	Common-Active PWM Model in CCM (a) direct equation-to-circuit	
	model (b) transformer model	33
3.9	Equivalent Circuit for Small-Signal PWM Switch in DCM $\ . \ . \ .$.	39
3.10	Equivalent Small-Signal Boost Circuit in CCM	40
3.11	Equivalent Small-Signal Boost Circuit in CCM After Transformation	
	Given in [21]	41

3.12	Equivalent Small-Signal Boost Circuit in CCM with Non-Idealities In-	
	cluded	
3.13	Bode Plot for Ideal Boost in CCM with $L = 180 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz$, and $\frac{V_O}{V_S} = 2.5$.	
3.14	Bode Plot for Non-Ideal Boost in CCM with $L = 180 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz, \frac{V_O}{V_S} = 2.5, r_L = 0.8 \Omega, \text{and} r_C = 0.1 \Omega. \dots \dots \dots$	
3.15	Step Response for Ideal Boost in CCM with $L = 180 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz$, and $\frac{V_O}{V_S} = 2.5$.	
3.16	Step Response for Non-Ideal Boost in CCM with $L = 180 \mu H, C =$	
	4.7 μF , $F_S = 350 kHz$, $\frac{V_O}{V_S} = 2.5$, $r_L = 0.8 \Omega$, and $r_C = 0.1 \Omega$	
3.17	Equivalent Small-Signal Boost Circuit in DCM	
3.18	Equivalent Small-Signal Non-Ideal Boost Circuit in DCM	
3.19	Bode Plot for Ideal Boost in DCM with $L = 18 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz$, and $\frac{V_O}{V_S} = 1.6.$	
3.20	Bode Plot for Non-Ideal Boost in DCM with $L = 18 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz, \frac{V_O}{V_S} = 1.6, r_L = 0.8 \Omega, \text{and} r_C = 0.1 \Omega.$	
3.21	Step Response for Ideal Boost in DCM with $L = 18 \mu H$, $C = 4.7 \mu F$,	
	$F_S = 350 kHz$, and $\frac{V_O}{V_S} = 1.6.$	
3.22	Step Response for Non-Ideal Boost in DCM with $L = 18 \mu H$, $C =$	
	4.7 μF , $F_S = 350 kHz$, $\frac{V_O}{V_S} = 1.6$, $r_L = 0.8 \Omega$, and $r_C = 0.1 \Omega$	
3.23	Open-Loop Gain Comparison for Varying Load Resistances with $L =$	
	$18 \mu H, C = 4.7 \mu F, F_S = 350 kHz, \text{and} M = 1.6. \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $	
3.24	Open-Loop Gain DCM Approximation Error	
3.25	Resonant Frequency Comparison for Varying Load Resistances with	
	$L = 18 \mu H, C = 4.7 \mu F, F_S = 350 kHz, \text{ and } M = 1.6. \dots \dots$	
3.26	Resonant Frequency DCM Approximation Error	
3.27	Quality Factor Comparison for Varying Load Resistances with $L =$	
	$18 \mu H, C = 4.7 \mu F, F_S = 350 kHz$, and $M = 1.6.$	
3.28	Quality Factor DCM Approximation Error	
3.29	RHP Zero Comparison for Varying Load Resistances with $L = 18 \mu H$,	
	$C = 4.7 \mu F, F_S = 350 kHz$, and $M = 1.6$	
3.30	RHP Zero DCM Approximation Error	

3.31	Pole Zero plot for Non-Ideal CCM and DCM Transfer Functions (RH-	
	PZs Not Plotted)	60
4.1	Boost Converter with Current Mode Control (Adapted from [29])	63
4.2	Circuit Implementation of a Voltage-Mode Controlled Boost Converter	63
4.3	Lag Compensator Circuit	65
4.4	Bode Plot for Lag Compensator Circuit with $R_O = 6 k\Omega$, $C_Z = 100 nF$,	
	$g_m = -1 S, R_T = 2 M\Omega, R_B = 500 k\Omega$ and R_Z swept with values of	
	100Ω , 500Ω , and $3 k\Omega$.	66
4.5	Lag Compensator Circuit	67
4.6	Bode Plot for Lag Compensator Circuit Plus Pole with $R_O = 6 k\Omega$,	
	$C_Z = 100 nF, C_C = 1 nF, g_m = -1 S, R_T = 2 M\Omega, R_B = 500 k\Omega$ and	
	R_Z swept with values of 100Ω , 500Ω , and $1 k \Omega$.	68
4.7	Lag Compensator Circuit with Feedback Zero	69
4.8	Bode Plot for Lag Compensator Circuit Plus Pole with $R_O = 6 k\Omega$,	
	$C_Z = 250 fF, C_C = 1 nF, g_m = -1 S, R_T = 2 M\Omega, R_B = 500 k\Omega$ and	
	R_Z swept with values of 40.7 $M\Omega$, 218 $M\Omega$, and 480 $M\Omega$	69
4.9	Lead Compensator Circuit	70
4.10	Bode Plot for Lead Compensator Circuit with $R_O = 6 k\Omega$, $C_1 = 30 pF$,	
	$C_C = 10 pF, g_m = -1 S, R_B = 500 k\Omega$, and R_T swept with values of	
	$2 M\Omega$, $8 M\Omega$, and $24 M\Omega$.	71
4.11	Lag-Lead Compensator Circuit	72
4.12	Bode Plot for Lag-Lead Compensator Circuit with $C_1 = 0.8 pF$, $C_C =$	
	$2 nF$, $g_m = -10 \mu S$, $R_B = 500 k\Omega$, $R_T = 2 M\Omega$, $C_Z = 300 nF$, and R_Z	
	swept with values of 100Ω , 500Ω , and $1 k \Omega$.	73
4.13	Pole-Zero Plot for Lag-Lead Compensator Circuit with $C_1 = 0.8 pF$,	
	$C_C = 2 nF, g_m = -10 \mu S, R_B = 500 k\Omega, R_T = 2 M\Omega, C_Z = 300 nF,$	
	and R_Z swept with values of 100Ω , 500Ω , and $1 k \Omega$	73
4.14	Bode Plot for Lag Compensator with DCM Boost with $G_{d0} = 5, C_O =$	
	$10\mu F,R_{ESR} = 0.5\Omega,Q = 0.05,\omega_0 = 16kHz,R_O = 3k\Omega,R_Z = 100\Omega,$	
	$C_Z = 10 \mu F, g_m = -1S, R_T = 2 M\Omega, \text{and} R_B = 500 k\Omega \ldots \ldots \ldots$	75

4.15	Bode Plot for Lag Compensator Plus Pole with DCM Boost with $G_{d0} =$	
	5, $C_O = 10 \mu F$, $R_{ESR} = 0.5 \Omega$, $Q = 0.05$, $f_0 = 16 kHz$, $R_O = 3 k\Omega$,	
	$R_Z = 100 \Omega, C_Z = 10 \mu F, C_C = 50 nF, g_m = -1S, R_T = 2 M\Omega$, and	
	$R_B = 500 k\Omega \qquad \dots \qquad $	76
5.1	Boost Converter with PID Control	78
5.2	CCM Boost Converter Bode Plot	79
5.3	CCM Boost Converter Pole-Zero Plot	79
5.4	CCM Boost Converter Bode Plot with PID Control; $g_m = 1.03 \mu S$,	
	$R_B = 560 k\Omega, R_T = 2 M\Omega, R_Z = 200 k\Omega, C_C = 13 pF, C_Z = 260 pF,$	
	$C_1 = 30 pF. \dots \dots \dots \dots \dots \dots \dots \dots \dots $	81
5.5	CCM Boost Converter Root-Locus with PID Control	81
5.6	CCM Boost Converter Step-Response with PID Control	82
5.7	CCM Boost Converter Phase Margin Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [150 mA, 300 mA, 450 mA], L = [170 \mu H,$	
	$200\mu H,230\mu H],g_m = [0.93\mu S,1.03\mu S,1.13\mu S].\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots$	83
5.8	CCM Boost Converter Bandwidth Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [150 mA, 300 mA, 450 mA], L = [170 \mu H,$	
	$200\mu H,230\mu H],g_m = [0.93\mu S,1.03\mu S,1.13\mu S].\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots$	84
5.9	CCM Boost Converter Gain Margin Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [150 mA, 300 mA, 450 mA], L = [170 \mu H,$	
	$200\mu H,230\mu H],g_m = [0.93\mu S,1.03\mu S,1.13\mu S].\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots\ldots$	84
5.10	CCM Boost Converter DC Gain Over Parameter Corners. $V_S = [1.8V,$	
	$3V, 4.2V$], $I_O = [150 mA, 300 mA, 450 mA$], $L = [170 \mu H, 200 \mu H,$	
	$230\mu H],g_m = [0.93\mu S,1.03\mu S,1.13\mu S].$	85
5.11	Boost Converter with PI Control	86
5.12	DCM Boost Converter Bode Plot	87
5.13	DCM Boost Converter Pole-Zero Plot (RHPZ Not Shown) \hdots	87
5.14	DCM Boost Converter with PI Control Bode Plot	89
5.15	DCM Boost Converter with PI Control Root Locus	89
5.16	DCM Boost Converter with PI Control Step Response	90

5.17	DCM Boost Converter Phase Margin Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H,$	
	$25\mu H$], $g_m = [8mS, 10mS, 12mS]$	91
5.18	DCM Boost Converter Bandwidth Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H,$	
	$25 \mu H$], $g_m = [8 mS, 10 mS, 12 mS]$	91
5.19	DCM Boost Converter Gain Margin Over Parameter Corners. $V_S =$	
	$[1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H,$	
	$25 \mu H$], $q_m = [8 mS, 10 mS, 12 mS$]	92
5.20	DCM Boost Converter DC Gain Over Parameter Corners. $V_S = [1.8V]$	
	$3V, 4.2V$], $I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H, 25 \mu H],$	
	$q_m = [8 mS, 10 mS, 12 mS].$	92
6.1	DCM Boost Converter Load Step from $10 mA$ to $20 mA$ with an Un-	
	derdamped Step Response	95
6.2	DCM Boost Converter Load Step from $10 mA$ to $20 mA$ with an Over-	
	damped Step Response	96
6.3	Voltage Injection via Opening the Loop (Adapted from [32])	98
6.4	Voltage Injection via Middlebrook's Method	98
6.5	Current Injection via Opening the Loop (Adapted from [32])	99
6.6	Current Injection via Middlebrook's Method	99
6.7	Voltage Injection Test Setup	100
6.8	8-bit PRBS Circuit.	102
6.9	8-bit PRBS Signal.	102
6.10	Autocorrelation of (a) Single-Period 8-bit PRBS, and (b) Four-Period	
	8-bit PRBS.	103
6.11	Injection Point for PRBS in a Switched DC-DC Converter	104
A.1	Block Diagram for a Queen-Bee Genetic Algorithm for Optimization	
	of Switched-Converter Controller Transfer Function	110
A.2	Comparison of QBGA Convergence with a Varied and Constant Rate	
	of Mutation	112

A.3	Comparison of Convergence Speeds for a) QBGA with NPFF and b)	
	QBGA with Linear Fitness.	114
A.4	Step Responses of QBGA with NPFF for a) Generation 1 and b) Gen-	
	eration 10	115
A.5	Step Response PSO at a) Iteration 1 and b) Iteration 40	117
A.6	Fitness Value Convergence Comparisons for QBGA and PSO	118
A.7	Step Response Comparisons Between a) QBGA and b) PSO for Full	
	System	118
B.1	Powerlib Toolbar	120
B.2	Series RLC Branch	120
B.3	Changing Series RLC Branch Parameters	121
B.4	Boost Converter in Powerlib (no Controller)	121
B.5	Boost Converter in Powerlib	122
B.6	Boost Converter in Powerlib Output Measurement	123
\mathbf{D}		

List of Tables

3.1	Comparison of Transfer Function Open Loop Gain Equations	54
3.2	Comparison of Transfer Function Resonant Frequency Equations	56
3.3	Comparison of Transfer Function Quality Factor Equations	58
3.4	Comparison of Transfer Function Zero Frequency Equations \ldots .	58
5.1	CCM Boost Converter DC Specs	78
5.2	CCM Boost Converter Pole and Zero Locations	78
5.3	CCM Boost Converter Parameter Corners	82
5.4	DCM Boost Converter DC Specs	86
5.5	DCM Boost Converter Pole and Zero Locations	86
5.6	DCM Boost Converter Parameter Corners	90

List of Notations

ΔV_O	Output Ripple Voltage
η	Efficiency
ω_0	Resonant Frequency in radians per second
ϕ_M	Phase Margin
$ au_L$	Unitless Time Constant Equivalent to $\frac{LF_S}{R}$
ζ	Damping Coefficient
C_D	Capacitance of Schottky Diode
C_{OSS}	Capacitance of MOS Switch
C_{SW}	Total Capacitance of Switching Elements
$D \text{ or } \hat{d}$	Duty Cycle
F_S	Switching Frequency
g_m	Transconductance of an Operational Transconductance Amplifier (OTA)
G_{d0}	Low Frequency Converter Gain
I_D	Diode Current
I_L or $\hat{i_L}$	Inductor Current
I_O	Output Current

M	Converter DC Gain
P_{SW}	Power Absorbed by All Switching Elements
Q	Quality Factor of Filter
R	Load Resistance
r_C	Capacitor Series Resistance
r_L	Inductor Series Resistance
r_{SW}	Parasitic Switching Resistance
T_S	Switching Period
V_c	Control Voltage (or Error Voltage)
V_O or $\hat{v_o}$	Output Voltage
V_S	Input Voltage
BIST	Built-In Self-Test
CCM	Continuous Conduction Mode
CMC	Current Mode Control
DCM	Discontinuous Conduction Mode
EMI	Electromagnetic Interference
ESR	Equivalent Series Resistor
FPGA	Field-Programmable Gate-Array
FRA	Frequency Response Analyzer
GA	Genetic Algorithm
KCL	Kirchoff's Current Law
KVL	Kirchoff's Voltage Law

LTI	Linear Time-Invariant
NPFF	Normalized Parabolic Fitness Function
OTA	Operational Transconductance Amplifier
PID	Proportional-Integral-Derivative
PRBS	Psuedo-Random Binary Sequence
PSO	Particle Swarm Optimization
PWM	Pulse-Width Modulation
QBGA	Queen-Bee Genetic Algorithm
RHP	Right-Hand-Side of Imaginary Plane
SMU	Source Measurement Unit
SSA	State Space Averaging
VMC	Voltage Mode Control

Chapter 1

Introduction

Modern mobile integrated circuits contain many different sub-circuits with varying degrees of power requirements [1]. Likewise, these circuit are typically designed to be high-performance and thus require fast and stable power supplies in order to guarantee proper operation [2]. Because of these requirements, Switched DC-DC Boost Converters have become commonplace in such integrated circuits due to their ability to 'boost' the small voltage of a battery coupled with their high efficiency [3–6].

In order to guarantee the commercial viability of a Boost Converter, the circuit itself must be well understood. This includes understanding the impact of non-ideal components on the DC and transient responses of the system, as well as understanding the regions that a converter may operate in. Understanding these effects helps to illustrate different regions that may cause instability in a DC-DC Boost Converter and, thus, gives important information to the designer as to what type of component tolerances and values are acceptable.

1.1 Importance of Stability

For low-performance circuits, converter stability may not be as important as other design criteria such as output voltage range or output power dissipation capabilities; however, for mobile integrated circuits, especially those used for display purposes, stability is a very important factor [7].

Maintaining a highly stable power supply for a display, such as an LCD, helps

to guarantee that pixel brightness exhibits little variation from the expected ideal value. This is a very important design consideration since the human eye perceives brightness on a non-linear scale, known as luminance. This means that there exist levels at which the change in voltage or current is very slight, yet the change in luminance is large. As such, it is important for a mobile-integrated-display-circuit power-supply to be very stable as any ringing could cause perceivable changes in pixel brightness which lowers the quality of the product [8].

An unfortunate reality for integrated display-circuits is that they typically have to function properly for a multitude of different display backplanes which, in turn, alters the load requirements for the circuit power supply [9]. In addition, these displays can have vastly different numbers of pixels which, yet again, alters the load requirements for the supply. In every one of these conditions, the need for stability (due to the luminance concerns addressed earlier) is of the utmost importance. Thus, the ability to accurately measure stability as well as understanding what areas of the converter are likely to cause *in*stability is paramount to ensuring a commercially viable product.

1.2 Thesis Organization

To begin the discussion of Boost Converter Stability, the DC performance is analyzed and discussed in Chapter 2. Two different Boost Converter operating modes are explored, known as Continuous-Conduction Mode (CCM), and Discontinuous-Conduction Mode. Each mode of operation is explored in terms of voltage conversion ratios, duty-cycle requirements, efficiency and output ripple. These results are then compared to determine the trade-offs of operation in either region. The efficiency of the converter is analyzed in terms of sensitivity to two primary sources of non-ideal power-loss in order to develop an understanding of their effects on DC performance.

Chapter 3 covers the small-signal analysis of a Boost Converter in order to develop accurate models that help to form an understanding of what conditions typically cause instability. Again, both the CCM and DCM converters are explored. First, a smallsignal modeling technique known as State-Space Averaging is explored wherein the Boost Converter is averaged and linearized around a given operating point [14]. As a comparison, a technique known as Averaged PWM Modeling is performed which produces similar results with less overall work [18, 19]. The transfer functions for both Boost Converters with ideal components and Boost Converters with non-ideal components are then compared via Bode plots and transient step responses. An approximation is presented wherein an non-ideal Boost Converter operating in DCM can be approximated to that of an ideal Boost Converter operating in DCM, which enables the simplification of various small-signal parameters.

Chapter 4 analyzes various error-amplifier control architectures for integrated circuits. These circuits utilize Operational Transconductance Amplifiers (OTAs) due to high speed and low layout area cost. Various control schemes are analyzed and compared via frequency response characteristics. Controller selection recommendations are made based on converter requirements. Appendix A also explores error-amplifier controller design via sophisticated high-dimensional optimization algorithms, specifically that of Genetic Algorithms and Particle Swarm Optimization Algorithms.

Chapter 5 deals with the full Boost Converter circuit from power conversion stage all the way through the control stage. Here, various design methods are presented based on converter region of operation (CCM or DCM) and the issues associated with ensuring a stable converter explored. The different architectures are compared via Bode plots and transient step response plots in order to better visualize the important differences. The converters are also analyzed via the complex plane in order to better visualize pole-zero placement and the effects on converter stability.

Chapter 6 introduces various Switched DC-DC Converter stability measurement techniques. These include ubiquitous step response plots as well as Bode plots. Another measurement technique explored utilizes the cross-correlation of input noise and the output waveform in order to extract a Bode plot mathematically. These stability test methods are explored and potential test implementations are presented. Recommendations are made based on the benefits of each test setup.

Chapter 2

Boost Converter DC Characteristics

A boost converter (shown in Figure 2.1) has two primary modes of operation: Continuous Conduction Mode (CCM) and Discontinuous Conduction Mode (DCM). As these names would indicate, they are defined by whether or not the converter is continuously or discontinuously conducting; specifically, whether or not the inductor current saturates or drops to zero within one switching period. This is explained graphically in Figure 2.2. In this chapter, the DC equations for voltage transfer function, output ripple voltage, and efficiency will be derived for both CCM and DCM. These equations will be compared in order to better determine the benefits of using either conduction method.



Figure 2.1: Ideal Boost Converter Circuit



Figure 2.2: Inductor Current in Boost Converter for CCM and DCM

2.1 Continuous Conduction Mode (CCM)

In order to begin analysis, it is important to observe the converter behavior without any added parasitics (switch resistance, inductor resistance, diode forward voltage drop, *et cetera*). Thus, Figure 2.1 will be used for the ideal derivations.

2.1.1 Ideal Converter

During steady-state operation when the switch is closed, the voltage across the capacitor causes the diode to become reverse-biased since the anode will be grounded. Thus, the voltage across the inductor, V_L is equal to the source voltage, V_S . Since the voltage across an inductor is equal to $L\frac{di_L}{dt}$ and the inductor current is approximately constant (in that it is increasing linearly), it can be said that (taking $dt = DT_S$):

$$\Delta i_{L,closed} = \frac{V_S D}{LF_S} \tag{2.1}$$

Next, it is necessary to open the switch. Since current through an inductor cannot change instantaneously and the diode is the only device which can provide a conduction path for the inductor current, the diode *must* become forward-biased. As such, the voltage across the inductor becomes $V_S - V_O$. Since the switch is open, and referring to the inductor current graph for CCM shown in Figure 2.2, $dt = (1-D)T_S$. Rearranging the equation for voltage in an inductor yields:

$$\Delta i_{L,open} = \frac{(V_S - V_O)(1 - D)}{LF_S}$$
(2.2)

In steady-state, the total change in inductor current must equal zero, thus by using (2.1) and (2.2):

$$0 = \Delta i_{L,closed} + \Delta i_{L,open}$$

$$0 = \frac{V_S D}{LF_S} + \frac{(V_S - V_O)(1 - D)}{LF_S}$$

$$0 = V_S D + V_S (1 - D) - V_O (1 - D)$$

Rearranging the above equation to solve for the DC output-to-input transfer function yields:

$$\frac{V_O}{V_S} = \frac{1}{1 - D} \tag{2.3}$$

Output Ripple

To solve for the output ripple voltage, it is important to consider how the charge is changing within the output filter capacitance. Consider the period for which the switch is closed, that is during the interval DT_S . Since the diode is reverse biased, the capacitor discharges in order to provide current to the load resistance. Thus, since the voltage at the output is positive and maintaining the convention that current flows into positive nodes,

$$I_C = -I_R = -I_O$$

Thus, for this interval the change in capacitor charge is given by the product of the current and the interval, or:

$$\Delta Q = -I_O DT = -\frac{I_O D}{F_S}$$

Because the charge in a capacitor is equal to the product of both the voltage and the capacitance, it can be said that:

$$\Delta Q = \Delta V C$$
$$\Delta V = \frac{\Delta Q}{C}$$

which yields the output ripple voltage equation of:

$$\Delta V_O = \frac{DI_O}{CF_S} \tag{2.4}$$

Efficiency

To find efficiency, a simple relation between input and output power can be established. Since there are no loss mechanisms in the circuit using the ideal diode model, all power delivered by the source will be absorbed in the load. Therefore the efficiency will be 100% for an ideal boost converter.

2.1.2 Non-Ideal Converter

Figure 2.3 shows the boost converter circuit with parasitics added in.

In order to derive the output voltage of the non-ideal boost converter in CCM, the concept of energy-conservation [10] is used wherein:

$$P_S = P_O + P_{Loss}$$

Thus it is necessary to observe what mechanisms will cause power loss in the non-ideal

model. Unfortunately, Figure 2.3 doesn't provide a completely accurate picture as to what will cause power loss as it ignores the dynamic switching losses in both the transistor switch and the diode. Since both the switch and diode have parasitic capacitances associated with them, during each switching cycle they need to be charged and discharged. Doing so provides a momentary short to ground and there will be non-negligible power loss as a result, given by (2.5).

$$P_{C_{SW}} = \frac{1}{2} C_{SW} V_{sw}^2 F_S \tag{2.5}$$

In the worst-case situation, the voltage across the switch, V_{sw} , will be equal to the output voltage plus at least the forward voltage drop of the diode, or $V_O + V_D$. The parameter C_{SW} is device-specific but typically has a range between $10 \, pF$ to $200 \, pF$ (commonly shown as C_{OSS} on datasheets). Figure 2.4 shows a plot for the normalized power consumption of all parasitic components.

There is one noticeable characteristic of Figure 2.4: operation in DCM is far more dependent on the dynamic switching loss than operation in CCM. The explanation is rather simple: in CCM, the current through the inductor never goes to zero, thus during the period in which the switch is open there is a continuous amount of power loss through the inductor and diode which ends up contributing to far more loss than the switching elements. In DCM, however, there is a period in which no current flows (Figure 2.2) and thus the power losses through the inductor and diode are much smaller in value. Because of this, the loss due to the switch capacitance becomes far more noticeable. Note that during the portion of the time there is no current flow in



Figure 2.3: Non-Ideal Boost Converter Circuit



Figure 2.4: Normalized Power Loss in CCM and DCM

the DCM operating mode, the switch capacitance will charge to V_S . This is because the diode must stop conducting which means there is no path for current from the input source, V_S , so the node on the positive side of the diode will charge to V_S . Over time, the total voltage change at the switch will be approximately equal to V_O rather than V_S and this will cause a discontinuity at the CCM/DCM boundary for efficiency calculations [11].

Now, consider the silicon diode as it begins to switch from reverse-biased to forward-biased. Here, the depletion region begins to decrease in width which, in turn, increases the depletion capacitance. However, there is also an increase in diffusion capacitance due to the minority charge located near the junction [12]. This diffusion capacitance will contribute to dynamic switching losses similar to that of the transistor. As a result, silicon diodes are commonly replaced by Schottky diodes in switched power systems because Schottky diodes do not store minority charge and, hence, do not exhibit diffusion capacitance [12]. The only capacitance they will have is due to the depletion region and will be quite small in comparison to the capacitance of a silicon diode, for example. This is an added benefit of using Schottky diodes in switched DC-DC converter circuits as increasing that capacitance causes a corresponding linear increase in power consumption as evidenced by plugging (2.6) into (2.5).

$$C_{SW} = C_{OSS} + C_D \tag{2.6}$$

Assuming that the diode ON-resistance is small and that the forward voltage drop will be the primary source of static loss in the diode (which is valid except for large load currents in CCM operation), the total power loss can be given as,

$$P_{Loss} = P_{r_L} + P_{SW} + P_D$$

where P_{SW} is equal to both the conduction loss due to the switch, $P_{r_{SW}}$, as well as the dynamic switching losses, $P_{C_{SW}}$ and P_D is equal to the conduction loss due to the diode.

Since the current through the inductor is always I_L by definition, the power absorbed by the inductor's series resistance is simply $r_L I_L^2$. Since the switch resistance will only absorb power when the switch is on, that is during time interval DT, and the current through it will be equivalent to I_L , it's clear that the static power absorbed by the switch will be $Dr_{SW}I_L^2$ whereas the dynamic power consumption is $\frac{1}{2}C_{SW}V_O^2F_S$. Likewise, since the diode is only conducting for the time interval of $(1 - D)T_S$, the average power absorbed by the diode is just $(1 - D)V_DI_L$. This yields the equation:

$$P_{Loss} = r_L I_L^2 + Dr_{SW} I_L^2 + \frac{1}{2} C_{SW} V_O^2 F_S + (1 - D) V_D I_L$$
(2.7)

Since,

$$P_{S} = P_{O} + P_{r_{L}} + P_{SW} + P_{D}$$

$$\therefore V_{S}I_{S} = V_{O}I_{O} + r_{L}I_{L}^{2} + Dr_{SW}I_{L}^{2} + \frac{1}{2}C_{SW}V_{O}^{2}F_{S} + (1-D)V_{D}I_{L}$$
(2.8)

Since current through the diode is equal to $(1-D)I_L$, and all of the DC component of the diode current must be delivered to the load, the following relationship can be established and then solved for I_L :

$$(1-D)I_L = I_O$$

$$I_L = \frac{I_O}{(1-D)}$$
(2.9)

Dividing (2.8) through by I_L and plugging (2.9) into I_L yields:

$$V_S = (1-D)V_O + r_L \frac{I_O}{1-D} + Dr_{SW} \frac{I_O}{1-D} + \frac{1-D}{I_O} C_{SW} V_O^2 F_S + (1-D)V_D$$

Plugging $\frac{V_O}{R}$ in for I_O and solving for V_O then results in the following expression for output voltage of a Boost Converter in CCM with parasitics:

$$V_O\left[(1-D) + \frac{r_L}{R}\frac{1}{1-D} + \frac{R(1-D)}{2}C_{SW}F_S + \frac{D}{1-D}\frac{r_{SW}}{R}\right] = V_S - (1-D)V_D$$
$$V_O = \frac{V_SR(1-D) - V_DR(1-D)^2}{R(1-D)^2 + r_L + Dr_{SW} + \frac{1}{2}(1-D)^2R^2C_{SW}F_S}$$
(2.10)

To check this answer, simply set all of the parasitics equal to zero and it's clear that (2.10) will yield the ideal expression shown in (2.3).

Output Ripple

The ripple voltage at the output due to the capacitor is the same in the non-ideal case as it is in the ideal case which is given in (2.4). However, there is an additional ripple voltage caused by the capacitor's series resistance. When the capacitor is charging, the current will be decreasing at the same rate as the changing inductor current, Δi_L , thus the largest peak current is $I_{L,max}$ which can be expressed as a sum of the average inductor current (2.9) and $\frac{\Delta i_L}{2}$.

First, an expression for Δi_L needs to be derived for the non-ideal converter. The process is similar to that of the ideal case wherein the voltage across the inductor is found and then solved for Δi_L . Referring to Figure 2.3, when the switch is closed it's clear that:

$$V_L = V_S - \Delta i_L (r_L + r_{SW})$$
$$L \frac{\Delta i_L}{DT_S} = V_S - \Delta i_L (r_L + r_{SW})$$

Rearranging and solving for Δi_L yields (2.11).

$$\Delta i_L = \frac{DV_S}{LF_S + D(r_L + r_{SW})} \tag{2.11}$$

Since I_L is given in (2.9), $I_{L,max}$ can be found as per (2.12).

$$I_{L,max} = I_L + \frac{\Delta i_L}{2}$$
$$I_{L,max} = \frac{I_O}{(1-D)} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}$$
(2.12)

Since the ripple voltage due to the capacitor's ESR can be expressed in (2.13) and $\Delta i_C = I_{L,max}$, (2.12) can be plugged into (2.13) for Δi_C to yield (2.14).

$$\Delta V_{O,esr} = \Delta i_C r_C \tag{2.13}$$

$$\Delta V_{O,esr} = \left[\frac{I_O}{(1-D)} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}\right] r_C$$
(2.14)

In a worst-case situation, the ripple peaks due to the capacitor and ESR coincide and thus will need to be summed in order to produce a correct expression for output voltage ripple. In realistic circuits, these peaks will rarely be at the exact same spot, so (2.15) is a worst-case expression for total output voltage ripple.

$$\Delta V_{O,total} = \frac{DI_O}{CF_S} + \left[\frac{I_O}{(1-D)} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}\right] r_C$$
(2.15)

Efficiency

The efficiency in a boost converter can be modeled as shown in (2.16). Here, the power loss, P_{Loss} , has already been derived in (2.7) where I_L is given in (2.9). Using these two relationships, it can be shown that P_{Loss} is equivalent to the expression in (2.17).

$$\eta = \frac{P_O}{P_O + P_{Loss}} \tag{2.16}$$

$$P_{Loss} = I_O \left[\frac{r_L I_O + D r_{SW} I_O + V_D (1 - D)^2}{(1 - D)^2} \right] + \frac{1}{2} C_{SW} V_O^2 F_S$$
(2.17)

Since P_O is simply the power absorbed by the load, the efficiency of the non-ideal converter in CCM can be expressed as (2.18) after plugging in for I_O .

$$\eta_{CCM} = \frac{R(1-D)^2}{R(1-D)^2 + r_L + Dr_{SW} + \frac{1}{2}C_{SW}F_SR(1-D)^2 + \frac{V_D}{V_O}R(1-D)^2}$$
(2.18)

2.2 Discontinuous Conduction Mode (DCM)

Similar to the process involved in deriving the DC equations for CCM, the ideal DCM converter will be considered before addition of parasitics. From a theoretical standpoint, the derivation is nearly identical to that of CCM with the exception of needing to find an expression for D_1 in place of (1 - D) (refer to Figure 2.2).

2.2.1 Ideal Converter

When the switch is closed in the circuit shown earlier in Figure 2.1, the same thing happens in DCM as CCM so $\Delta i_{L,closed}$ is the same as (2.1). When the switch is open, however, there is a slight difference due to the fact that the inductor current goes to zero before the switching cycle. This difference can be attained by simply replacing the (1 - D) term in (2.2) with D_1 , as shown in (2.19).

$$\Delta i_{L,open} = \frac{(V_S - V_O)D_1}{LF_S} \tag{2.19}$$

Since the sum of the change in inductor currents over one period must equal zero, adding (2.1) with (2.19) and solving for V_O results in an ideal DC output voltage equation shown in (2.20).

$$V_O = V_S \left(\frac{D+D_1}{D_1}\right) \tag{2.20}$$

However, (2.20) doesn't really say much since D_1 is currently unknown. Solving for this is important and also fairly easy; since the Δi_L is zero before any switching cycle starts (as this is what defines DCM operation), the maximum current, and thus the height of the triangular waveform in Figure 2.2, can be given by $\Delta i_{L,closed}$ in (2.1). From here, consider the average current through the diode. Since the diode only conducts during the D_1 cycle (as it is reverse-biased when the switch is closed as described in Section 2.1.1), the average current through it is given by:

$$I_{D,avg} = \frac{1}{T_S} \left(\frac{1}{2} I_{max} D_1 T_S \right)$$
$$I_{D,avg} = \frac{1}{2} I_{max} D_1$$

When the diode is conducting, all of its current is delivered to the load, such that,

$$I_{D,avg} = I_O = \frac{V_O}{R}$$

After substituting (2.1) in for I_{max} and solving for D_1 yields (2.21):

$$D_1 = I_O\left(\frac{2LF_S}{V_SD}\right) = \left(\frac{V_O}{V_S}\right)\left(\frac{2LF_S}{RD}\right)$$
(2.21)

From here, equation (2.21) can be plugged into (2.20),

$$\frac{V_O}{V_S} = \frac{D + \begin{pmatrix} V_O \\ V_S \end{pmatrix} \left(\frac{2LF_S}{RD}\right)}{\begin{pmatrix} V_O \\ V_S \end{pmatrix} \left(\frac{2LF_S}{RD}\right)}$$
$$0 = \left(\frac{V_O}{V_S}\right)^2 \left(\frac{2LF_S}{RD}\right) - \left(\frac{V_O}{V_S}\right) \left(\frac{2LF_S}{RD}\right) - D$$
$$\therefore 0 = \left(\frac{V_O}{V_S}\right)^2 - \left(\frac{V_O}{V_S}\right) - \frac{RD^2}{2LF_S}$$

Solving this quadratic for $\frac{V_O}{V_S}$ yields the final DCM output votlage expression shown below in (2.22).

$$V_{O} = \frac{V_{S}}{2} \left(1 + \sqrt{1 + \frac{2D^{2}}{\tau_{L}}} \right)$$
(2.22)

where

$$\tau_L = \frac{LF_S}{R} \tag{2.23}$$

Output Ripple

The derivation for output ripple is the same as that of CCM since the output current waveform is identical for the period when the switch is on. Because of this, the change in charge in the output filter capacitor is $-I_O DT_S$, thus the output ripple for both CCM and DCM is given by (2.4).

Efficiency

As was the case for CCM, all of the input power must be delivered to the load due to the absence of loss mechanisms in the ideal DCM converter. Thus, efficiency must be 100%.

2.2.2 Non-Ideal Converter

In the Non-Ideal CCM converter, power loss was observed in order to obtain an output voltage expression and the same technique is used here for the Non-Ideal DCM Converter. The same parasitic losses in CCM operation contribute loss in DCM such that,

$$P_{Loss} = P_{r_L} + P_{SW} + P_D$$
$$P_S = P_O + P_{Loss}$$

Unlike the CCM derivation, however, it is not accurate to simply state that $P_S = V_S I_S$. In DCM, and as Figure 2.2 shows, there is a period of time during each switching cycle that the inductor current goes to zero. Since the switch is open, current can only flow from the source, through the inductor and diode, and then into the load. However, since there is no current in the inductor, there cannot be any current from the source which implies no power is being supplied by the source during that period. Thus, it's clear that,

$$P_S = (D + D_1)V_S I_S$$

Again, since the inductor only conducts during DT_S and D_1T_S , its series resistance only absorbs power during that time. The rest of the loss mechanisms are similar to that of the CCM converter where the switch resistance absorbs power during DT_S while the diode only absorbs during D_1T_S . Using this knowledge, (2.24) can be constructed in order to derive an expression for the output voltage.

$$(D+D_1)V_SI_L = V_OI_O + (D+D_1)r_LI_L^2 + Dr_{SW}I_L^2 + \frac{1}{2}C_{SW}V_O^2F_S + D_1V_DI_L \quad (2.24)$$

Since the output current is the same as the inductor current during the D_1 cycle, I_O can be said to be the same as D_1I_L . Solving for I_L and plugging into (2.24) yields:

$$DV_S + D_1 V_S = D_1 V_O + \frac{D}{D_1} r_L I_O + r_L I_O + \frac{D}{D_1} r_{SW} I_O + \frac{D_1}{2I_O} C_{SW} V_O^2 F_S + D_1 V_D$$

From here, a few re-definitions are needed in order to keep the resulting equation manageable. First, recalling that D_1 is given as $\left(\frac{2LF_s}{DR}\right) \left(\frac{V_O}{V_S}\right)$ in (2.21) and $\tau_L = \frac{LF_S}{R}$ in (2.23), redefine:

$$M = \frac{V_O}{V_S} \tag{2.25}$$

$$r_X = \frac{r_L + r_{SW}}{R} \tag{2.26}$$

$$\kappa = \frac{V_D}{V_S} \tag{2.27}$$

$$\tau_{sw} = RC_{SW}F_S \tag{2.28}$$

Thus, D_1 becomes $\frac{2\tau_L}{D}M$ and the above parameters can be plugged into the power conservation equation which results in,

$$M^{2} \left[1 + \frac{\tau_{sw}}{2} \right] + M \left[\frac{Dr_{L}}{2R\tau_{L}} + \kappa - 1 \right] + \left[\frac{D^{2}(Dr_{X} - 2\tau_{L})}{(2\tau_{L})^{2}} \right] = 0$$

Solving this quadratic equation for M and plugging $\frac{V_O}{V_S}$ back into M yields the final output voltage equation for a Non-Ideal converter in DCM as shown in (2.29).

$$V_{O} = V_{S} \left[\frac{\sqrt{2D^{2}(\tau_{sw}+2)(2\tau_{L}-r_{X})+4\tau_{L}(\kappa-1)^{2}\left[\tau_{L}+D\frac{r_{L}}{R}\right]+\left(D\frac{r_{L}}{R}\right)^{2}}{2\tau_{L}(\tau_{sw}+2)} \cdots \frac{2\tau_{L}(\kappa-1)-\frac{r_{L}}{R}}{2\tau_{L}(\tau_{sw}+2)} \right]$$
(2.29)

Obviously, (2.29) is rather vague due to the re-definitions from (2.25)-(2.28), but

it lends itself well to solving via software such as MATLAB[®]. In order to check the answer, eliminate all of the parasitics (κ , r_X , τ_{sw} , and r_L). This process yields:

$$M = \frac{\sqrt{8\tau_L D^2 + 4\tau_L^2} + 2\tau_L}{4\tau_L}$$
$$V_O = \frac{V_S}{2} \left(1 + \sqrt{\frac{8\tau_L D^2}{4\tau_L^2} + \frac{4\tau_L^2}{4\tau_L^2}} \right)$$
$$V_O = \frac{V_S}{2} \left(1 + \sqrt{1 + \frac{2D^2}{\tau_L}} \right) \checkmark$$

which verifies the ideal converter equation in (2.22).

Output Ripple

Just like in the ideal case, the output ripple due to the capacitor is identical to that of a converter operating in CCM (2.4). The ripple due to the ESR, however, does have a slight change: (1 - D) in equation (2.12) is replaced with D_1 . Plugging in for D_1 with (2.21) yields the following new expression for $I_{L,max}$ in DCM:

$$I_{L,max} = \frac{DV_S}{2LF_S} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}$$
(2.30)

Thus, by utilizing the expression for the ESR ripple voltage found previously in (2.13), the ESR ripple in DCM is determined by (2.31) where the worst-case total output ripple voltage is given by (2.32).

$$\Delta V_{O,esr} = \left[\frac{DV_S}{2LF_S} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}\right] r_C \tag{2.31}$$

$$\Delta V_{O,total} = \frac{DI_O}{CF_S} + \left[\frac{DV_S}{2LF_S} + \frac{DV_S}{2LF_S + 2D(r_L + r_{SW})}\right] r_C \tag{2.32}$$

Efficiency

To find the efficiency of the Non-Ideal converter in DCM, equation (2.16) is employed where P_{Loss} is given by (2.33).

$$P_{Loss} = (D + D_1)r_L I_L^2 + Dr_{SW} I_L^2 + \frac{1}{2}C_{SW} V_O^2 F_S + D_1 V_D I_L$$

$$P_{Loss} = \frac{D}{D_1^2} r_L I_O^2 + \frac{1}{D_1} r_L I_O^2 + \frac{D}{D_1^2} r_{SW} I_O^2 + \frac{1}{2} C_{SW} V_O^2 F_S + V_D I_O$$

$$P_{Loss} = \frac{V_O^2}{R} \left[\left(\frac{D}{2LF_S} \right)^2 \left(\frac{V_S}{V_O} \right)^2 R D r_L + \frac{Dr_L}{2LF_S} \left(\frac{V_S}{V_O} \right) \cdots + \left(\frac{D}{2LF_S} \right)^2 \left(\frac{V_S}{V_O} \right)^2 D r_{SW} + \frac{1}{2} R C_{SW} F_S + \frac{V_D}{V_O} \right]$$
(2.33)

Plugging (2.33) into (2.16) results in an expression for the efficiency of a Non-Ideal converter in DCM given in (2.34).

$$\eta_{DCM} = \frac{1}{1 + \frac{D^3 R}{(2LF_S)^2} \left(\frac{V_S}{V_O}\right)^2 [r_L + r_{SW}] + \frac{Dr_L}{2LF_S} \left(\frac{V_S}{V_O}\right) + \frac{1}{2}RC_{SW}F_S + \frac{V_D}{V_O}}$$
(2.34)

2.3 CCM vs. DCM

Being able to compare the different parameters for CCM and DCM is important to develop understanding of the design trade-offs involved. However, comparison isn't as straight-forward as it initially may seem since operation in either mode depends on a number of parameters (L, F_S , I_O , etc.). In order to properly compare the two modes, the *realistic* operation needs to be compared to the converter ONLY in CCM and the converter ONLY in DCM (despite there being regions where the equations are invalid). To do so, the boundary between CCM and DCM needs to be defined.

Refer to Figure 2.2. Since DCM operation is defined by the inductor current reaching zero before the end of the switching cycle, the transition between CCM and DCM must occur when $D + D_1 = 1$. Thus,

$$1 = D + \left(\frac{V_O}{V_S}\right) \left(\frac{2LF_S}{RD}\right)$$
$$L = \frac{R(D - D^2)V_S}{2F_S V_O}$$

Since $\frac{V_O}{V_S} = \frac{1}{1-D}$ given in (2.3):

$$L_{crit} = \frac{(1 - \frac{V_S}{V_O})V_S^2}{2F_S V_O I_O}$$
(2.35)


Figure 2.5: Output Voltage versus Duty Cycle Comparison for DCM and CCM

So if $L > L_{crit}$, that implies that the converter is operating in CCM whereas in any other case the converter is operating in DCM. Using this definition allows relatively seamless transition between modes, as shown in Figure 2.5. Note that the following parameters were used in all examples in this section: $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $V_S = 3.5 V$, $F_S = 350 \,kHz$, $r_L = 1 \,\Omega$, $r_{SW} = 0.3 \,\Omega$, $r_C = 0.1 \,\Omega$, $V_D = 0.3 \,V$, $C_{OSS} = 50 \,pF$, and $C_D = 10 \,pF$ with an ideal $V_O = 5.5 \,V$ and I_O varying as $1 \,mA$, $15 \,mA$, and $35 \,mA$. The load resistance, R, was simply calculated by taking the ideal V_O value and dividing by each defined load current.

Figures 2.6 and 2.7 show the converter efficiency and ripple voltage. As mentioned in Section 2.1.2, the discontinuity in the efficiency plot can be attributed to the approximation that in DCM the switch capacitance charges and discharges to V_O rather than V_S [11].

Sensitivity to Parasitics

Observing the sensitivity that different parasitics have on the DC operation of the converter is important to develop an understanding of which parameters contribute most to the non-ideal behavior. Figures 2.8, and 2.9 depict the sensitivity of the converter efficiency in both CCM and DCM with respect to r_L and C_{SW} , respectively.

The inductor resistance, r_L exhibits similar sensitivity effects for both CCM and



Figure 2.6: Efficiency versus Output Current Comparison for DCM and CCM



Figure 2.7: Output Ripple Voltage versus Duty Cycle Comparison for DCM and CCM

DCM with the DCM plot being slightly more affected. For large resistances (10 Ω or higher), the sensitivity of the efficiency plots approaches -1 which implies that an r_L of those sizes would essentially render the converter unusable (since efficiency, at best, can be 1). Typical inductor series resistance values fall in the $m\Omega$ range, and thus have a much smaller impact on efficiency (as indicated in Figure 2.8). However, the slope of the sensitivity in this range is quite large, thus any small variations in r_L can



Figure 2.8: Sensitivity of Efficiency to Changing r_L for CCM and DCM



Figure 2.9: Sensitivity of Efficiency to Changing C_{SW} for CCM and DCM

result in a rather substantial decrease in efficiency. Another problem associated with this is that typically an inductor's series resistance will have a positive temperature coefficient so as the inductor's temperature begins to increase, the resistance will increase, and thus the efficiency will decrease. The equation for this sensitivity plot is found in (2.36) and (2.37).

$$S_{r_L}^{\eta_{CCM}} = \frac{-r_L}{R(1-D)^2 + r_L + Dr_{SW} + \frac{1}{2}C_{SW}F_SR(1-D)^2 + \frac{V_D}{V_O}R(1-D)^2}$$
(2.36)

$$S_{r_{L}}^{\eta_{DCM}} = \frac{-r_{L} \left[\frac{D^{3}R}{(2LF_{S})^{2}} \left(\frac{V_{S}}{V_{O}} \right)^{2} + \frac{DR}{2LF_{S}} \left(\frac{V_{S}}{V_{O}} \right) \right]}{1 + \frac{D^{3}R}{(2LF_{S})^{2}} \left(\frac{V_{S}}{V_{O}} \right)^{2} [r_{L} + r_{SW}] + \frac{Dr_{L}}{2LF_{S}} \left(\frac{V_{S}}{V_{O}} \right) + \frac{1}{2}RC_{SW}F_{S} + \frac{V_{D}}{V_{O}}}$$
(2.37)

The plot in Figure 2.9 is almost comical in that the converter in CCM seems to be totally unaffected by increasing the C_{SW} value, whereas the converter in DCM has a nearly linear relationship with increasing this value. It should be noted that if the graph were to be extrapolated into the μF range that the curve resembled something more along the lines of Figure 2.8. For switch capacitance values less than 1 nF, the sensitivity of the DCM converter will increase by 0.005 for roughly every 100 pFincrease in capacitance. The sensitivity equations for CCM and DCM are given in (2.38) and (2.39), respectively.

Overall, the capacitance has a significantly smaller impact on the efficiency than that of the inductor series resistance. It may not be a negligible amount, but it's fairly clear the inductor series resistance poses a very large threat to converter operation.

$$S_{C_{SW}}^{\eta_{CCM}} = \frac{-\frac{1}{2}RC_{SW}F_S(1-D)^2}{R(1-D)^2 + r_L + Dr_{SW} + \frac{1}{2}C_{SW}F_SR(1-D)^2 + \frac{V_D}{V_O}R(1-D)^2}$$
(2.38)

$$S_{C_{SW}}^{\eta_{DCM}} = \frac{-\frac{1}{2}RC_{SW}F_S}{1 + \frac{D^3R}{(2LF_S)^2} \left(\frac{V_S}{V_O}\right)^2 [r_L + r_{SW}] + \frac{Dr_L}{2LF_S} \left(\frac{V_S}{V_O}\right) + \frac{1}{2}RC_{SW}F_S + \frac{V_D}{V_O}}$$
(2.39)

Final Thoughts

It's rather difficult to state which mode is "better" since each have their own merits. Operating in DCM allows for the inductor size to be smaller since it doesn't need to maintain a current through each switching cycle. However, DCM is affected far more by the dynamic switching power loss mechanisms as discussed previously and as evidenced by Figure 2.6. As Figure 2.5 shows, as the output current decreases, the dynamic switching loss begins to have a large impact on the converter efficiency and is much worse in the case of the DCM converter. However, despite the apparent shortcomings of operating a Boost Converter in DCM, there is increased controllability which is important to maintain a stiff, regulated output. This will be discussed in detail throughout Chapter 3.

Chapter 3

Small-Signal Analysis

A conventional DC switched converter consists of three main components: DC Converter, Error Amplifier, and PWM Generator. This architecture is shown in block diagram form in Figure 3.1.

The DC characteristics of the converter block have already been discussed in Chapter 2. The purpose of this block is simply to take a DC voltage, V_S , and convert it to a different, scaled, voltage. This block can assume a number of different architectures (Buck, Boost, Buck-Boost, *etc.*), but the boost architecture is what is specifically focused on within the contents of this thesis. That said, Figure 3.1 is a general form that can be used for any DC-DC converter.

The Error Amplifier block is simply an amplifier that serves to minimize the steady-state error of the converter. This is typically done with a resistive voltage divider that is fed from the output of the Converter block to the input of an Op-Amp or OTA. The various Error Amplifier architectures are explored more thoroughly throughout Chapter 4.



Figure 3.1: Regulated DC Converter Block Diagram (Adapted from [13])

Finally, the last major block within a regulated DC-DC converter is the PWM Generator. This takes the error signal produced by the amplifier, V_c , and converts it to a digital signal in order to properly drive the switch within the converter. Based on the DC level of V_c , the PWM Generator will modulate the pulse-width which results in a varying duty-cycle.

Understanding the role of each of these blocks is incredibly important because the system is, inherently, a feedback network. This requires careful consideration to be taken in regards to small-signal performance has the system as the potential to become unstable and create an oscillator instead of a steady, regulated output voltage.

This chapter will illustrate the analytic techniques used for deriving small-signal models and transfer functions for a Boost Converter in CCM and DCM operation. The PWM Generator block transfer function will also be derived. Since the Error Amplifier is observed in more detail within Chapter 4, the small-signal analysis of that block will be deferred.

3.1 PWM Generation

Figure 3.2 depicts the conventional circuit used for PWM generation. The amplified error signal, V_c , is fed into the inverting terminal on the comparator and a voltage ramp operating at the switching frequency, F_S , is fed into the non-inverting terminal. When the error voltage matches the value of the input ramp, the comparator triggers



Figure 3.2: PWM Generation Circuit



Figure 3.3: PWM Generation Circuit Waveforms

high (causing the switch to open) which, with a varying error signal, will cause a modulation of the duty cycle in the output digital waveform. Modulation of this duty cycle is what controls the output voltage of the converter block. This process is depicted schematically in Figure 3.3.

Since the input to the PWM block is V_c and the output is d, it is necessary to find the small-signal transfer function $\frac{d}{V_c}$. Doing so is rather straight forward by observation of the waveforms in Figure 3.3: the duty cycle will simply be equal to the ratio of the error signal to the ramp voltage peak (as V_c increases, the duty cycle follows linearly). Thus,

$$P(s) = \frac{d}{V_c} = \frac{1}{V_{peak}} \tag{3.1}$$

3.2 State-Space Averaging

The idea of using State-Space Averaging (commonly shortened to SSA) as a way to model switched converters was first introduced by R.D. Middlebrook and Slobodan Ćuk [14]. Since then, it has widely been used as a way to extract various small-signal and averaged DC transfer functions [15–17].

The basic premise is simple: by obtaining *averaged* values for each relevant node voltage and current, a model can be developed which effectively ignores the effects of the high-frequency switching which, in turn, allows for a linear small-signal model to be developed. In order to properly obtain a small-signal model for a given converter, the following steps must be performed:

- 1. Obtain equations for each state (closed/open).
- 2. Average the equations over one switching period.
- 3. Perturb the average equations so that each variable has a DC and AC component.
- 4. Take Laplace Transform.
- 5. Place new equations into matrices.
- 6. Extract desired transfer function.

To illustrate this process, an ideal Boost Converter operating in CCM will be used.

State 1: Switch Closed

The first state under consideration where the switch is closed is shown schematically in Figure 3.4. Using KVL along the left-hand loop, it's clear that:

$$V_{S} = V_{L}$$

$$V_{S} = L \frac{di_{L}}{dt}$$

$$\therefore \frac{di_{L}}{dt} = \frac{V_{S}}{L}$$
(3.2)

Applying KCL along the right-hand loop yields:

$$I_C = -I_R$$

$$C\frac{dv_o}{dt} = -\frac{v_o}{R}$$

$$\therefore \frac{dv_o}{dt} = -\frac{v_o}{RC}$$
(3.3)

State 2: Switch Open

The next state is when the switch is open, as shown schematically in Figure 3.5. Using KVL along the left-hand loop, and KCL along the right-hand loop (as was done previously), it's clear that:

$$V_{S} = V_{L} + v_{o}$$

$$V_{S} = L \frac{di_{L}}{dt} + v_{o}$$

$$\therefore \frac{di_{L}}{dt} = \frac{V_{S}}{L} - \frac{v_{o}}{L}$$
(3.4)



Figure 3.4: Ideal Boost Converter with Switch Closed



Figure 3.5: Ideal Boost Converter with Switch Open

And,

$$I_C = i_L - I_R$$

$$C \frac{dv_o}{dt} = i_L - \frac{v_o}{R}$$

$$\therefore \frac{dv_o}{dt} = \frac{i_L}{C} - \frac{v_o}{RC}$$
(3.5)

Averaging the State Equations

Because the converter is assumed to be operating in CCM, the switch will be closed for the duration of some time dT_S while it will be open for some time $(1 - d)T_S$ (where d is the duty-cycle and T_S is the switching period). Thus, the averaged state equations can be determined by adding (3.2) with (3.4) and (3.3) with (3.5) such that:

$$\dot{i_L} = \frac{1}{T_S} \left[dT_S \left(\frac{V_S}{L} \right) + (1 - d) T_S \left(\frac{V_S}{L} - \frac{v_o}{L} \right) \right]$$
$$\dot{i_L} = \frac{1}{L} V_S - \frac{1 - d}{L} v_o$$
(3.6)

And,

$$\dot{v_o} = \frac{1}{T_S} \left[dT_S \left(-\frac{v_o}{RC} \right) + (1-d) T_S \left(\frac{i_L}{C} - \frac{v_o}{RC} \right) \right]$$
$$\dot{v_o} = \frac{1-d}{C} i_L - \frac{1}{RC} v_o \tag{3.7}$$

Perturbation and Laplace Transformation

In order to derive small-signal models, the averaged equations must contain small signal variables. This is done by replacing each variable with both a DC and AC value such that a variable x would be equated to $X + \hat{x}$. Performing this perturbation to the variables i_L , v_o , and d yields:

$$\frac{d(I_L + \hat{i_L})}{dt} = \frac{1}{L}V_S - \frac{1 - D - \hat{d}}{L}(V_O + \hat{v_o})$$
$$\frac{d(V_O + \hat{v_o})}{dt} = \frac{1 - D - \hat{d}}{C}(I_L + \hat{i_L}) - \frac{1}{RC}(V_O + \hat{v_o})$$

After expanding the above equations and eliminating both pure DC quantities and quantities containing multiples of AC quantities (since two small numbers multiplied by each other yield an even smaller number), the following equations are produced:

$$\hat{i}_{L} = \frac{\hat{d}}{L} V_{O} - \frac{1}{L} \hat{v}_{o} + \frac{D}{L} \hat{v}_{o}$$
(3.8)

$$\hat{v_o} = \frac{1 - D}{C}\hat{i_L} - \frac{\hat{d}}{C}I_L - \frac{1}{RC}\hat{v_o}$$
(3.9)

Taking the Laplace Transform of (3.8) and (3.9) and rearranging such that the state variables \hat{i}_L and \hat{v}_o end up on the right-hand side of each equation yield:

$$\hat{d}V_O = sL\hat{i}_L + (1-D)\hat{v}_o \tag{3.10}$$

$$\hat{d}I_L = (1-D)\hat{i}_L - \left(sC + \frac{1}{R}\right)\hat{v}_o$$
 (3.11)

Matrix Creation

Creating the state matrices is straight forward by observation of (3.10) and (3.11):

$$\begin{bmatrix} V_O \\ I_L \end{bmatrix} \hat{d} = \begin{bmatrix} sL & (1-D) \\ (1-D) & -\left(sC + \frac{1}{R}\right) \end{bmatrix} \begin{bmatrix} \hat{i_L} \\ \hat{v_o} \end{bmatrix}$$

Since the goal is to obtain a transfer function for $\frac{\hat{v}_{\alpha}}{\hat{d}}$, the inverse of the A-Matrix (that is, the 2 × 2 matrix on the right-hand side of the above equation) must be pre-multiplied on each side and the variable \hat{d} must be divided by each side such that the expression in (3.12) results,

$$\frac{1}{\hat{d}} \begin{bmatrix} \hat{i}_L \\ \hat{v}_o \end{bmatrix} = \begin{bmatrix} sL & (1-D) \\ (1-D) & -\left(sC + \frac{1}{R}\right) \end{bmatrix}^{-1} \begin{bmatrix} V_O \\ I_L \end{bmatrix}$$
(3.12)

Taking the inverse of the A matrix can be tedious, especially as the matrix dimen-

sions begin to increase. As such, it's wise to use a program such as MATLAB[®] in order to eliminate any potential mistakes that are likely to arise in a hand-derivation. The inverse of the A-matrix is shown in 3.13. Thus, it's clear that the expression in 3.14 is the correct small-signal control-to-output transfer function.

$$A^{-1} = \frac{1}{s^2 R L C + sL + R(1-D)^2} \begin{bmatrix} sRC + 1 & R(1-D) \\ R(1-D) & -sLR \end{bmatrix}$$
(3.13)

$$\frac{\hat{v_o}}{\hat{d}} = \frac{V_O}{1-D} \left[\frac{-sL + R(1-D)^2}{s^2 R L C + sL + R(1-D)^2} \right]$$
(3.14)

In a more general form, this transfer function can be expressed as:

$$\frac{\hat{v_o}}{\hat{d}} = G_{d0} \frac{(1 - \frac{s}{\omega_{z,1}})(1 + \frac{s}{\omega_{z,2}})}{\frac{s^2}{\omega_0^2} + \frac{s}{Q\omega_0} + 1}$$
(3.15)

$$G_{d0} = \frac{V_O}{1 - D}$$
(3.16)

$$\omega_{z,1} = \frac{R(1-D)^2}{L} \tag{3.17}$$

$$\omega_{z,2} = \infty \tag{3.18}$$

$$\omega_0 = \frac{1 - D}{\sqrt{LC}} \tag{3.19}$$

$$Q = R(1-D)\sqrt{\frac{C}{L}}$$
(3.20)

SSA is a very nice way to generate switched-converter transfer functions; however, they're quite tedious. Any time the converter topology changes or components are added to the circuit (parasitics, for example), all of the state equations and perturbations need to be repeated.

3.3 Average PWM Model

As a solution to the problems associated with State-Space Averaging, Vatché Vorpérian introduced the idea of an averaged PWM Switch model [18,19]. Since the only switch-

ing components within a converter are the diode and switch itself, Vorpérian reasoned that an equivalent circuit model for this switch/diode combination would be simpler than SSA as a single, general model could be generated and placed in any converter topology [18].

The circuit used to generated the equivalent circuit for both CCM and DCM operation is shown in Figure 3.6. There are three terminals: active, passive, and common. The active node is the node which connects to only the switch, the passive connects only to the diode, and the common connects to both (as Figure 3.6 indicates). Once the equivalent PWM Switch circuit is derived, it can be used in *any* configuration in *any* converter topology; in essence, it only needs to be derived once.

The PWM Switch can be derived in three different ways, depending on the orientation of the switch. Regardless of which configuration used, the resultant circuit model will be the same. Since the analysis up to this point has focused solely on boost converter topologies, the switch will be assumed to be in a common-active configuration for all derivations, as shown in Figure 3.7. This nomenclature refers to which node is connected to the common point on the circuit, typically ground. Thus, three orientations can be used: common-active, common-passive, or common-common.

3.3.1 PWM Switch in CCM

The analysis of the PWM Switch in CCM is rather straight forward under the assumption that the current through the inductor is constant. Consider the active current, I_a . When the switch is on, during time interval dT_S , $I_a = I_c$. When the switch is



Figure 3.6: PWM Switch (adapted from [18] and [19])



Figure 3.7: Ideal Boost Converter with Common-Active PWM Switch



Figure 3.8: Common-Active PWM Model in CCM (a) direct equation-to-circuit model (b) transformer model

closed, the active node is floating and, thus, $I_a = 0$. Averaging these conditions over the entire switching period, T_S , results in (3.21). Likewise, during the interval dT_S , the common-passive voltage, V_{cp} will be the same as the active-passive voltage, V_{ap} since in this state $V_c = V_a$. When the switch is opened, $V_c = V_p$, thus, $V_{cp} = 0$. Averaging V_{cp} over the entire switching period results in (3.22).

$$I_a = dI_c \tag{3.21}$$

$$V_{cp} = dV_{ap} \tag{3.22}$$

These equations are then perturbed, similar to the process used in the State-Space Averaging technique explained earlier. After perturbation of the variables I_a , d, V_{cp} , and V_{ap} (along with the cancellation of higher-order small-signal terms), the small-signal parameters of \hat{i}_a and \hat{v}_{cp} can be derived, as shown in (3.23) and (3.24), respectively.

$$\hat{i_a} = D\hat{i_c} + \hat{d}I_C \tag{3.23}$$

$$\hat{v_{cp}} = D\hat{v_{ap}} + \hat{d}V_{AP} \tag{3.24}$$

Figure 3.8a shows the average PWM Model in CCM with (3.23) and (3.24) dropped right into the circuit while Figure 3.8b shows the circuit with a 1 : D ideal transformer used to replace the $D\hat{ic}$ voltage source and $D\hat{v_{ap}}$ current source.

3.3.2 PWM Switch in DCM

To begin analysis, it is necessary to consider both switching events; that is when the switch is open, dT_S , and when the switch is closed, d_1T_S . By defining the currents at each node of the PWM switch as always *entering* a node, the following can be said of the average voltages and currents during the "on" state:

$$I_{c} = -I_{a} = I_{L}$$

$$I_{L,avg} = \frac{I_{L,max}}{2}$$

$$\therefore I_{a} = -d\frac{I_{L,max}}{2}$$
(3.25)

Since the change in inductor current, Δi_L , in DCM will go from zero to some maximum value during one switching cycle, the change will just end up equaling that maximum value. Since $I_{L,max} = \Delta i_L$, it is clear that:

$$V_{ac} = -V_L = -L \frac{\partial i_L}{\partial t}$$
$$V_{ac} = -L \frac{\Delta i_L}{\Delta t}$$

$$V_{ac} = -L \frac{I_{L,max}}{dT_S} \tag{3.26}$$

During the "off" state, it is clear that the same process can be used to derive expressions for I_p and V_{cp} such that,

$$I_{c} = -I_{p} = I_{L}$$

$$I_{L,avg} = \frac{I_{L,max}}{2}$$

$$I_{p} = -d_{1}\frac{I_{L,max}}{2}$$
(3.27)

Likewise, since this is a small-signal derivation, $V_S = 0 V$ which implies:

$$V_{cp} = -V_L = -L \frac{\partial i_L}{\partial t}$$

$$V_{cp} = -L \frac{\Delta i_L}{\Delta t}$$

$$V_{cp} = -L \frac{I_{L,max}}{d_1 T_S}$$
(3.28)

From these equations, the following relationships between terminal currents and voltages can be derived:

$$-\frac{I_{L,max}}{2} = \frac{I_p}{d_1} = \frac{I_a}{d}$$

and

$$-L\frac{I_{L,max}}{T_S} = dV_{ac} = d_1 V_{cp}$$

Therefore, it's clear that:

$$I_a = I_p \frac{d}{d_1} \tag{3.29}$$

$$V_{ac} = V_{cp} \frac{d_1}{d} \tag{3.30}$$

Rearranging equations (3.25) and (3.26) and plugging it into (3.28) and (3.27),

respectively, yields

$$V_{cp} = \frac{2LI_a F_S}{(d)(d_1)}$$

and

$$V_{ac} = \frac{2LI_p F_S}{(d)(d_1)}$$

$$\therefore d_1 = \frac{2LF_S}{d} \frac{I_a}{V_{cp}} = \frac{2LF_S}{d} \frac{I_p}{V_{ac}}$$
(3.31)

This expression for d_1 can then be plugged into (3.30) to yield:

$$V_{ac} = \frac{2LF_S}{d} \frac{I_a}{V_{cp}} (\frac{1}{d}) V_{cp}$$
$$V_{ac} = \frac{2LF_S}{d^2} \frac{I_a}{V_{cp}} V_{cp}$$
$$\therefore V_{cp} = \frac{d^2}{2LF_S} \frac{V_{cp}}{I_a} V_{ac}$$

and

$$V_{ac} = \frac{2LF_S}{d} \frac{I_p}{V_{ac}} (\frac{1}{d}) V_{cp}$$
$$V_{ac} = \frac{2LF_S}{d^2} \frac{I_p}{V_{ac}} V_{cp}$$
$$\therefore V_{cp} = \frac{d^2}{2LF_S} \frac{V_{ac}}{I_p} V_{ac}$$

•

Thus,

$$V_{cp} = \mu V_{ac} \tag{3.32}$$

where

$$\mu = \frac{d^2}{2LF_S} \frac{V_{cp}}{I_a} = \frac{d^2}{2LF_S} \frac{V_{ac}}{I_p} = \frac{d}{d_1}$$
(3.33)

Now, by taking the expression for d_1 in (3.31) and substituting it into (3.29), the following relationship appears:

$$I_a = I_p d(\frac{d}{2LF_S})(\frac{I_a}{V_{cp}})$$

$$I_a = \frac{d^2}{2LF_S} (\frac{I_a}{V_{cp}}) I_p$$

$$\therefore I_a = \mu I_p \tag{3.34}$$

Using the average equations derived in the previous section, a small-signal model can be derived by perturbing each node current and voltage as well as the duty-cycle. To perform this perturbation analysis, the following relationships must be established:

$$I_a = I_A + \hat{i_a}$$
$$I_p = I_P + \hat{i_p}$$
$$V_{ac} = V_{AC} + \hat{v_{ac}}$$
$$d = D + \hat{d}$$

The first step will be to substitute these values, which contain steady-state and small-signal components, into both (3.34) and (3.31) (and ignoring higher-order small-signal quantities),

$$I_A + \hat{i_a} = \frac{D + \hat{d}}{d_1} (I_P + \hat{i_p})$$

$$d_1 = \frac{2LF_S}{D + \hat{d}} \frac{I_P + \hat{i_p}}{V_{AC} + \hat{v_{ac}}}$$

$$\therefore I_A + \hat{i_a} = \frac{D + \hat{d}}{\frac{2LF_S}{D + \hat{d}} \frac{I_P + \hat{i_p}}{V_{AC} + \hat{v_{ac}}}} (I_P + \hat{i_p})$$

$$I_A + \hat{i_a} = \frac{DV_{AC} + D\hat{v_{ac}} + \hat{d}V_{AC}}{2LF_S} (D + \hat{d})$$

$$\therefore \left(\frac{I_A}{V_{AC}}\right) \left(\frac{2LF_S}{D^2}\right) (I_A + \hat{i_a}) = I_A + \frac{I_A}{V_{AC}} \hat{v_{ac}} + \frac{2I_A}{D} \hat{d}$$

From (3.30), $V_{AC} = \frac{D_1}{D} V_{CP}$, therefore:

$$\left(\frac{1}{D_1}\right) \left(\frac{I_A}{V_{CP}}\right) \left(\frac{2LF_S}{D}\right) (I_A + \hat{i_a}) = I_A + \frac{I_A}{V_{AC}} \hat{v_{ac}} + \frac{2I_A}{D} \hat{d}$$

From (3.31), $D_1 = \frac{2LF_S}{D} \frac{I_A}{V_{CP}}$, so the term on the left-hand side of the equation cancels leaving:

$$\hat{i_a} = \frac{I_A}{V_{AC}}\hat{v_{ac}} + \frac{2I_A}{D^2}\hat{d}$$

From here, the coefficients to the small signal parameters can be defined as:

$$g_i = \frac{I_A}{V_{AC}} \tag{3.35}$$

$$k_i = \frac{2I_A}{D} \tag{3.36}$$

$$\therefore \hat{i_a} = g_i \hat{v_{ac}} + k_i \hat{d} \tag{3.37}$$

Next it is necessary to perturb equation (3.30) in order to obtain a relationship for the small-signal current at the passive terminal, $\hat{i_p}$:

$$\begin{split} V_{AC} + \hat{v_{ac}} &= \frac{d_1}{D + \hat{d}} (V_{CP} + \hat{v_{cp}}) \\ V_{AC} + \hat{v_{ac}} &= \frac{2LF_S}{D^2 + 2D\hat{d}} \left(\frac{I_P V_{CP} + I_P \hat{v_{cp}} + \hat{i_p} V_{CP}}{V_{AC} + \hat{v_{ac}}} \right) \\ \frac{V_{AC}^2 + 2V_{AC} \hat{v_{ac}}}{2LF_S} &= \frac{I_P V_{CP} + I_P \hat{v_{cp}} + \hat{i_p} V_{CP}}{D^2 + 2D\hat{d}} \\ (I_P + \hat{i_p}) V_{CP} &= \frac{V_{AC}^2 + 2V_{AC} \hat{v_{ac}}}{2LF_S} (D^2 + 2D\hat{d}) - I_P \hat{v_{cp}} \\ I_P + \hat{i_p} &= \frac{1}{2LF_S} \left[\frac{V_{AC}^2 D^2}{V_{CP}} + \frac{2V_{AC}^2 D\hat{d}}{V_{CP}} + \frac{2D^2 V_{AC} \hat{v_{ac}}}{V_{CP}} \right] - \frac{I_P}{V_{CP}} \hat{v_{cp}} \end{split}$$

From (3.30), $\frac{V_{AC}}{V_{CP}} = \frac{D_1}{D}$, therefore:

$$I_P + \hat{i_p} = \frac{D_1}{2LF_S} [DV_{AC} + 2V_{AC}\hat{d} + 2D\hat{v_{ac}}] - \frac{I_P}{V_{CP}}\hat{v_{cP}}$$
$$\therefore I_P + \hat{i_p} = \frac{I_P}{DV_{AC}} [DV_{AC} + 2V_{AC}\hat{d} + 2D\hat{v_{ac}}] - \frac{I_P}{V_{CP}}\hat{v_{cP}}$$
$$\hat{i_p} = \frac{2I_P}{D}\hat{d} + \frac{2I_P}{V_{AC}}\hat{v_{ac}} - \frac{I_P}{V_{CP}}\hat{v_{cp}}$$



Figure 3.9: Equivalent Circuit for Small-Signal PWM Switch in DCM

Thus, the following parameters can be defined:

$$k_0 = \frac{2I_P}{D} \tag{3.38}$$

$$g_0 = \frac{I_P}{V_{CP}} \tag{3.39}$$

$$g_f = \frac{2I_P}{V_{AC}} \tag{3.40}$$

which results in a final expression for the small-signal current at the passive node as:

$$\hat{i}_p = g_f \hat{v_{ac}} + k_0 \hat{d} - g_0 \hat{v_{cp}}$$
(3.41)

Now, since expressions for both the small-signal currents at the active and passive nodes have been derived, a small-signal circuit model can be created for the PWM switch as shown in Figure 3.9. Note that this model is in agreement with those derived by both by Vorpérian in [19] and by Reatti and Kazimierczuk in [20].

3.4 Transfer Functions

In order to derive the small-signal transfer functions for the boost converter in both CCM and DCM, the derived models (Figures 3.8 and 3.9, respectively) will be placed

into the boost architecture and then analyzed. Using State-Space Averaging, the ideal CCM transfer function was derived in (3.14) but will be re-derived in order to illustrate the accuracy of the PWM Model. Parasitics will be added to the circuit as well in order to observe their impact on small-signal performance (specifically, the inductor resistance and capacitor resistance).

3.4.1 CCM: Ideal Transfer Function

By taking the CCM small-signal model for the PWM Switch derived earlier and placing it into the boost converter circuit, Figure 3.10 can be created. This circuit is now a linear small-signal model which can be analyzed to find any given transfer function (control-to-output, input-to-output, etc). Since the control-to-output, $\frac{\hat{v}_o}{\hat{d}}$, is the transfer function of concern here, the small-signal source v_i is set to 0.

In order to simplify analysis, the circuit shown in Figure 3.10 can be modified, as explained in [21]. This simplification involves moving the voltage source and current source to the *other* side of the transformer and connecting the capacitor and load resistance to the common, as opposed to passive, terminal. The resultant circuit is shown in Figure 3.11.

From here, nodal analysis can be used to derive the proper transfer function. First, observe that $\hat{v}_p = \hat{v}_o$. Since $\hat{v}_{cp} = D\hat{v}_{ap}$, due to the transformer's 1 : D turns ratio, the value for \hat{v}_c can be derived as follows:

$$\hat{v_c} - \hat{v_o} = -D\hat{v_o}$$



Figure 3.10: Equivalent Small-Signal Boost Circuit in CCM



Figure 3.11: Equivalent Small-Signal Boost Circuit in CCM After Transformation Given in [21]

$$\hat{v}_c = \hat{v}_o (1 - D) \tag{3.42}$$

Using nodal analysis, and plugging V_O in for V_P and plugging $\frac{V_O}{R(1-D)}$ in for I_C results in the following expression,

$$\frac{-V_O}{R(1-D)^2}\hat{d} = \frac{\hat{v}_c - V_O\hat{d}}{sL} + \frac{sC\hat{v}_c}{(1-D)^2} + \frac{\hat{v}_c}{R(1-D)^2}$$
$$\frac{-V_O}{R(1-D)^2}\hat{d} = \frac{(1-D)}{sL}\hat{v}_o - \frac{V_O}{sL}\hat{d} + \frac{sC}{1-D}\hat{v}_o + \frac{1}{R(1-D)}\hat{v}_c$$

After rearranging and finding a common denominator, the following expression emerges,

$$\hat{v}_o \left[\frac{s^2 R L C + sL + R(1-D)^2}{sLR(1-D)} \right] = \hat{d} V_O \left[\frac{R(1-D)^2 - sL}{sLR(1-D)^2} \right]$$

thus, $\frac{\hat{v_o}}{\hat{d}}$ is given as:

$$\frac{\hat{v_o}}{\hat{d}} = \frac{V_O}{1-D} \left[\frac{-sL + R(1-D)^2}{s^2 RLC + sL + R(1-D)^2} \right] \checkmark$$

which verifies the answer found in (3.14).



Figure 3.12: Equivalent Small-Signal Boost Circuit in CCM with Non-Idealities Included

3.4.2 CCM: Non-Ideal Transfer Function

By adding in the non-ideal inductor and capacitor series resistances, as shown in Figure 3.12, a more accurate (and, unfortunately, more complex) transfer function can be derived. The same process used in the ideal derivation will be used here. Since $\hat{v}_c = \hat{v}_o(1-D)$ from the previous derivation, it's clear that:

$$-\frac{V_O}{R(1-D)^2}\hat{d} = \frac{(1-D)\hat{v_o} - V_O\hat{d}}{sL + r_L} + \frac{sC}{(sr_CC+1)(1-D)}\hat{v_o} + \frac{1}{R(1-D)}\hat{v_o}$$

Rearranging the above equation so that the \hat{v}_o and \hat{d} terms are separated and above common denominators yields:

$$\hat{v}_o \left[\frac{R(sr_CC+1)(1-D)^2 + s^2RLC + sRr_LC + (sL+r_L)(sr_CC+1)}{R(sL+r_L)(sr_CC+1)(1-D)} \right] \cdots$$
$$\cdots = \hat{d}V_O \left[\frac{R(1-D)^2 - (sL+r_L)}{R(sL+r_L)(1-D)^2} \right]$$

After solving for $\frac{\hat{v}_o}{\hat{d}}$ and expanding the terms in both the numerator and denominator, the small-signal non-ideal transfer function for a boost converter in CCM was found as shown in (3.43).

$$\frac{\hat{v_o}}{\hat{d}} = \frac{V_O}{1-D} \left[\frac{-s^2 \frac{r_C}{R} L + s(r_C(1-D)^2 - \frac{L}{RC} - \frac{r_C r_L}{R}) - \frac{r_L}{RC} + \frac{1}{C}(1-D)^2}{s^2 L(\frac{r_C}{R} + 1) + s[r_C(1-D)^2 + r_L + \frac{r_C r_L}{R} + \frac{L}{RC}] + \frac{r_L}{RC} + \frac{(1-D)^2}{C}} \right]$$
(3.43)



Figure 3.13: Bode Plot for Ideal Boost in CCM with $L = 180 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and $\frac{V_O}{V_S} = 2.5$.

Alternatively, the transfer function can be expressed in the more general form given in (3.15) with:

$$G_{d0} = \frac{V_O}{1 - D} \frac{R(1 - D)^2 - r_L}{R(1 - D)^2 + r_L}$$
(3.44)

$$\omega_{z,1} = \frac{R(1-D)^2 - r_L}{L} \tag{3.45}$$

$$\omega_{z,2} = \frac{1}{r_C C} \tag{3.46}$$

$$\omega_0 = \frac{1}{\sqrt{LC}} \sqrt{\frac{R(1-D)^2 + r_L}{R + r_C}}$$
(3.47)

$$Q = \sqrt{LC} \frac{\sqrt{(R+r_C)(R(1-D)^2 + r_L)}}{Rr_C C(1-D)^2 + Rr_L C + r_C r_L C + L}$$
(3.48)

Ideal and Non-Ideal Comparisons

Figure 3.13 shows the Bode plot for the ideal transfer function given in (3.14). Figure 3.14 shows the Bode plot for the non-ideal transfer function given in (3.43). The parameters used were: $L = 180 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 2.5$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.



Figure 3.14: Bode Plot for Non-Ideal Boost in CCM with $L = 180 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 2.5$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.

A notable feature of Figure 3.13 is the resonant peak that occurs for large load resistances. The explanation for this is rather simple: as the load resistance increases, the load current must decrease. Since the diode is reverse-biased during the interval when the switch is closed, the load current must be supplied by the capacitor. Since load current is small, the RC time constant is large indicating that it will take a large amount of time for the capacitor to fully discharge. Since this is the case, there is a large amount of energy still stored in the capacitor at the time where the switch opens. This, along with the energy stored in the inductor, causes resonance if the switch is modulated at an appropriate frequency. The location of this peak can easily be predicted by (3.19) and is 2.2 kHz for the ideal transfer. In the non-ideal case, the peak exhibits a slight decrease in frequency due to the series resistances of the inductor and capacitor as shown in (3.47).

Figures 3.15 and 3.16 depict the step responses for the ideal and non-ideal CCM converters, respectively. As predicted by the Bode plot comparison, the non-ideal converter is noticeably more stable. Likewise, stability begins to increase with an increase in load current.

From a transient perspective, it's clear that the loss mechanisms in the circuit contribute to a decrease in the steady-state value of the converter. This is expected



Figure 3.15: Step Response for Ideal Boost in CCM with $L = 180 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and $\frac{V_O}{V_S} = 2.5$.



Figure 3.16: Step Response for Non-Ideal Boost in CCM with $L = 180 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 2.5$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.

from the equations derived in Chapter 2 that showed that the ideal CCM converter has no load dependence in (2.3), while the non-ideal CCM converter *does* in (2.10). This indicates that a controller used to try and maintain a fixed DC voltage for varying loads not only has to compete with the inherent instability of the converter (thanks to the conjugate pair pole in the system), but also needs to provide enough gain in order to maintain a regulated output.

3.4.3 DCM: Ideal Transfer Function

Using the circuit derived in Figure 3.9 and placing it into the boost converter model results in the circuit shown in Figure 3.17.

In order to derive the control-to-output transfer function (ignoring all parasitics), the small-signal source \hat{v}_i needs to be set to zero. Doing so allows for the following equation to be developed using KCL at the control node:

$$0 = \frac{v_{ac}}{sL} + v_{ac}g_i + k_i\hat{d} + g_f v_{ac} - v_{cp}g_o + k_o\hat{d}$$

Since $v_a = 0$ and $v_p = \hat{v}_o$, with some rearranging of terms, the following expression for v_c can be generated:

$$v_c = \frac{\hat{v}_o s L g_o + d[s L(k_i + k_o)]}{s L(g_i + g_f + g_o) + 1}$$
(3.49)

Next, it is necessary to develop an expression using KCL at the output node:

$$0 = v_{cp}g_o - g_f v_{ac} - k_o \hat{d} - v_o sC - \frac{v_o}{R}$$

Again, with some rearranging and by substituting $v_a = 0$ and $v_p = \hat{v_o}$, another



Figure 3.17: Equivalent Small-Signal Boost Circuit in DCM

expression for v_c can be generated:

$$v_c = \frac{\hat{v_o}[sC + \frac{1}{R} + g_o] + k_o \hat{d}}{g_o + g_f}$$
(3.50)

In order to obtain an expression for the transfer function $\frac{\hat{v}_o}{\hat{d}}$, (3.49) must be set equal to (3.50):

$$\hat{v_o} \frac{sC + \frac{1}{R} + g_o}{g_o + g_f} + \hat{d} \frac{k_o}{g_o + g_f} = \hat{v_o} \frac{sLg_o}{sL(g_i + g_f + g_o) + 1} + \hat{d} \frac{sL(k_i + k_o)}{sL(g_i + g_f + g_o) + 1}$$
$$\hat{v_o} \left[\frac{sRC + Rg_o + 1}{R(g_o + g_f)} - \frac{sLg_o}{sL(g_i + g_o + g_f) + 1} \right] = \cdots$$

$$\cdots \hat{d} \left[\frac{SL\kappa_i + SL\kappa_o}{sL(g_i + g_o + g_f) + 1} - \frac{\kappa_o}{g_o + g_f} \right]$$

$$\hat{v}_{o} \left[\frac{(sRC + g_{o}R + 1)(sL[g_{i} + g_{f} + g_{o}] + 1) - sLRg_{o}(g_{o} + g_{f})}{R(g_{o} + g_{f})(sL[g_{i} + g_{f} + g_{o}] + 1)} \right] = \cdots$$
$$\cdots \hat{d} \left[\frac{sL(k_{i} + k_{o})(g_{o} + g_{f}) - sLk_{o}(g_{i} + g_{o} + g_{f}) - k_{o}}{(g_{o} + g_{f})(sL[g_{i} + g_{o} + g_{f}] + 1)} \right]$$

Therefore, the small-signal control-to-output transfer function is:

$$\frac{\hat{v_o}}{d} = \frac{sLR[k_i(g_f + g_o) - k_o g_i] - k_o R}{s^2 LRC[g_i + g_o + g_f] + s[RC + L(g_i + g_o + g_f) + LRg_o g_i] + g_o R + 1}$$
(3.51)

Equation (3.51) verifies the findings reported in [20] by Reatti and Kazimierczuk for the boost-converter control-to-output transfer function.

For a proper comparison, this transfer function can be manipulated in order to fit the general form of (3.15). First, recall the definitions for g_i , k_i , k_0 , g_0 , and g_f given in (3.35), (3.36), (3.38), (3.39), and (3.40), respectively. By analyzing these parameters and plugging in for known circuit quantities, the expressions can be manipulated as shown in (3.52)-(3.56) where $M = \frac{V_O}{V_S}$ and $\tau_L = \frac{LF_S}{R}$:

$$g_{i} = \frac{I_{A}}{V_{AC}} = \frac{\frac{D}{D_{1}}\frac{V_{O}}{R}}{D_{1}V_{O}} = \frac{D}{R} \left(\frac{D}{2\tau_{L}M}\right)^{2}$$
(3.52)

$$g_0 = \frac{I_P}{V_{CP}} = \frac{\frac{V_O}{R}}{DV_O} = \frac{1}{RD}$$
(3.53)

$$g_f = \frac{2I_P}{V_{AC}} = \frac{2\frac{V_O}{R}}{D_1 V_O} = \frac{D}{R} \frac{1}{M\tau_L}$$
(3.54)

$$k_{i} = \frac{2I_{A}}{D} = -\frac{2\frac{V_{O}}{R}}{D_{1}} = -\frac{D}{R}\frac{V_{O}}{M\tau_{L}}$$
(3.55)

$$k_o = \frac{2I_P}{D} = -\frac{2\frac{V_O}{R}}{D} = -\frac{2V_O}{RD}$$
(3.56)

After plugging the above equations into (3.51) and some algebraic manipulation, the following parameters were extracted for the general transfer function in (3.15).

$$G_{d0} = \frac{2V_O}{1+D}$$
(3.57)

$$\omega_{z,1} = -\frac{2(M\tau_L)^2}{D^2 \frac{L}{R} \left(\frac{D}{2} + \frac{M\tau_L}{D}\right)}$$
(3.58)

$$\omega_{z,2} = \infty \tag{3.59}$$

$$\omega_0 = \frac{2M\tau_L}{\sqrt{LC}} \sqrt{\frac{1+D}{D^4 + 4D^2M\tau_L + 4M^2\tau_L^2}}$$
(3.60)

$$Q = \frac{1}{\omega_0} \left[\frac{4(M\tau_L)^2 (1+D)}{4RCDM^2 \tau_L^2 + \frac{L}{R} (D^4 + 4D^2 M \tau_L + 4M^2 \tau_L^2) + \frac{L}{R} D^3} \right]$$
(3.61)

3.4.4 DCM: Non-Ideal Transfer Function

Using Figure 3.18, the non-ideal DCM small-signal transfer function can be derived. The process follows the same structure as the ideal derivation where nodal analysis was utilized.

First, observe the currents at the \hat{v}_c node. Using KCL, the following expression

can be created:

$$0 = \frac{\hat{v_c}}{sL + r_L} + \hat{v_c}g_i - k_i\hat{d} + \hat{v_{cp}}g_0 + g_f\hat{v_c} - k_0\hat{d}$$

Such that:

$$\hat{v}_c = \frac{\hat{v}_o g_0(sL + r_L) + \hat{d}(k_i + k_0)(sL + r_L)}{(sL + r_L)(g_i + g_0 + g_f) + 1}$$
(3.62)

Moving to the output of the converter and, again, utilizing KCL gives the following expression:

$$0 = \hat{v_{pc}g_0} + g_f \hat{v_{ac}} + k_0 \hat{d} + \frac{\hat{v_o}sC}{sr_C C + 1} + \frac{\hat{v_o}sC}{R}$$

After rearranging terms and plugging (3.62) in for \hat{v}_c yields:

$$\hat{v}_{o} \left[\frac{s(RCr_{C}g_{0} + RC + Cr_{C}) + Rg_{0} + 1}{sRCr_{C} + R} - \frac{g_{0}(sL + r_{L})(g_{f} + g_{0})}{(sL + r_{L})(g_{i} + g_{f} + g_{0}) + 1} \right] = \cdots$$
$$\cdots \hat{d} \left[\frac{(g_{f} + g_{0})(k_{i} + k_{0})(sL + r_{L}) - k_{0}(sL + r_{L})(g_{i} + g_{f} + g_{0}) - k_{0}}{(sL + r_{L})(g_{i} + g_{f} + g_{0}) + 1} \right]$$

Clearly, the resulting equation will be very long. Thus, in order to condense the transfer function, the form of (3.63) is used where each parameter is given in (3.64) through (3.69):

$$T(s) = \frac{a_2 s^2 + a_1 s + a_0}{b_2 s^2 + b_1 s + b_0}$$
(3.63)



Figure 3.18: Equivalent Small-Signal Non-Ideal Boost Circuit in DCM

$$a_2 = LRCr_C[(g_f + g_0)k_i - k_0g_i]$$
(3.64)

$$a_1 = RCr_C[r_L(g_f + g_0)k_i - r_Lk_0g_i - k_0]\cdots$$

$$\cdots + LR[(g_f + g_0)k_i - k_0g_i]$$
 (3.65)

$$a_0 = Rr_L[(g_f + g_0)k_i - k_0g_i] - Rk_0$$
(3.66)

$$b_2 = LRCr_C g_i g_0 + LC(R + r_C)(g_i + g_0 + g_f)$$
(3.67)

$$b_1 = LRg_0g_i + L(g_i + g_0 + g_f) + RCr_C(r_Lg_i + 1)\cdots$$

$$\cdots + (RC + r_CC)[r_L(g_i + g_0 + g_f) + 1]$$
(3.68)

$$b_0 = r_L(g_i + g_0 + g_f) + g_0 g_i R r_L + R g_0 + 1$$
(3.69)

These values verify the findings in [20]. Of course, to better compare to the previous functions, it's important to fit these values to the general equation given in (3.15). Obtaining ω_0 and Q are straight forward, but obtaining the zeros of the transfer function are not quite as simple. The best approach is to place the numerator into an A matrix and solve for the eigenvalues. The A matrix has a structure of:

$$A = \frac{1}{a_0} \begin{bmatrix} -a_1 & -a_2 \\ & & \\ a_0 & 0 \end{bmatrix}$$

By plugging (3.64)-(3.66) into the A matrix and solving for the eigenvalues, λ_i , the zeros of the function can be extrapolated as:

$$\omega_{z,1} = \frac{1}{\lambda_1}$$
 and $\omega_{z,2} = \frac{1}{\lambda_2}$

After performing the aforementioned operations, the general-structure parameters were found to be:

$$G_{d0} = \frac{\frac{2V_O}{D} + Rr_L \left(\frac{D^2 V_O}{2(M\tau_L)^2 R^2} - \frac{V_O}{R} - \frac{D^2 V_O}{RM\tau_L}\right)}{r_L \left(\frac{1}{DR} + \frac{D^3}{4(M\tau_L)^2 R} + \frac{D}{M\tau_L R}\right) + \frac{1}{D} + \frac{D^2 r_L}{4(M\tau_L)^2 R} + 1}$$
(3.70)

$$\omega_{z,1} = -\frac{r_L D^3 + 2r_L D M \tau_L - 4R (M \tau_L)^2}{L D^3 + 2M \tau_L L D}$$
(3.71)



Figure 3.19: Bode Plot for Ideal Boost in DCM with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and $\frac{V_O}{V_S} = 1.6$.

$$\omega_{z,2} = \frac{1}{r_C C} \tag{3.72}$$

$$\omega_0 = \frac{1}{\sqrt{LC}} \sqrt{\frac{(1+D)(r_L D^3 + 4R(M\tau_L)^2) + 4M\tau_L r_L (D^2 + M\tau_L)}{D^3 r_C + (R+r_C)(D^4 + 4D^2 M\tau_L + 4(M\tau_L)^2)}}$$
(3.73)

$$Q = \frac{r_L [D^2 (D^2 + 4M\tau_L) + 4(M\tau_L)^2 + D^3] + 4R(M\tau_L)^2(1+D)}{\omega_0 Q_d}$$
(3.74)

$$Q_d = L(D^4 + D^3 + 4M\tau_L D^2 + 4(M\tau_L)^2) + r_C C(r_L D^3 + 4R(M\tau_L)^2) \cdots$$

$$\cdots + (RC + rc_C)[r_L(D^4 + 4M\tau_L D^2 + 4(M\tau_L)^2) + 4DR(M\tau_L)^2]$$
(3.75)

Ideal and Non-Ideal Comparisons

A comparison of the Bode plots for the ideal and non-ideal DCM transfer functions is shown in Figure 3.19 and Figure 3.20, respectively. The parameters used were: $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 1.6$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.

Besides the noticeable effects due to the additional zero given in (3.72) which, ultimately, has little effect on the overall converter performance, there is a noticeable decrease in open-loop gain as the current increases. Once the load resistance drops below a certain range, 0.72Ω for the parameters listed above, the open-loop gain dips below 0 dB which renders the converter essentially inoperable (without external



Figure 3.20: Bode Plot for Non-Ideal Boost in DCM with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 1.6$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.



Figure 3.21: Step Response for Ideal Boost in DCM with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and $\frac{V_O}{V_S} = 1.6$.

control, that is).

Figures 3.21 and 3.22 show the step response for the ideal and non-ideal DCM transfer functions, respectively. Here, two different loads were used, as indicated on each figure, in order to see the transient behavior as a function of the load. The steady-state behavior is a function of the load, as predicted in the DC voltage gain function



Figure 3.22: Step Response for Non-Ideal Boost in DCM with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, $\frac{V_O}{V_S} = 1.6$, $r_L = 0.8 \,\Omega$, and $r_C = 0.1 \,\Omega$.

in (2.22) in Chapter 2. This is a primary reason why feedback control is incredibly important as the duty cycle (the only controllable parameter in the circuit) must be modulated in order to maintain a given steady-state DC voltage for varying loads. As shown previously, this is not the case for CCM; however, the two-pole behavior of the system requires control in order to move the 0 dB crossing point such that the converter is stable in all required operating conditions.

Another observation of these two step responses is that they look identical. This indicates that the ideal and non-ideal transfer function may be approximately equal which, in turn, also implies that the loss mechanisms present in the circuit have a negligible effect on transient and small-signal performance. This will be discussed more thoroughly throughout the next section.

3.5 Transfer Function Comparison

It's difficult to compare the transfer functions for CCM and DCM since both are defined by different operating regions. A function valid in DCM, by definition, cannot be valid in CCM. Various parameters have to be carefully selected to guarantee that the converter is operating in a desired region. This is why the CCM and DCM



Figure 3.23: Open-Loop Gain Comparison for Varying Load Resistances with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and M = 1.6.

converters were generated with different parameters.

To alleviate this problem, it's possible to observe the individual transfer function parameters as a point of comparison (referencing the general transfer function equation given in (3.15)). Table 3.1 shows a comparison of the Open-Loop Gain expressions for each transfer function while Figure 3.23 shows a comparison of the open-loop gain values for varying load resistance. Here, the only parameter being modified to achieve the correct region of operation is the duty cycle, D.

A very interesting feature of Figure 3.23 is that the open-loop gain for the converter in DCM is very similar in both the ideal and non-ideal case across a large range of load

 Table 3.1: Comparison of Transfer Function Open Loop Gain Equations

	G_{d0}
CCM (Ideal)	$\frac{V_O}{1-D}$
DCM (Ideal)	$\frac{2V_O}{1+D}$
CCM (Non-Ideal)	$\frac{V_O}{1-D} \frac{R(1-D)^2 - r_L}{R(1-D)^2 + r_L}$
DCM (Non-Ideal)	$\frac{\frac{2V_O}{D} + Rr_L \left(\frac{D^2 V_O}{2(M\tau_L)^2 R^2} - \frac{V_O}{R} - \frac{D^2 V_O}{RM\tau_L}\right)}{r_L \left(\frac{1}{DR} + \frac{D^3}{4(M\tau_L)^2 R} + \frac{D}{M\tau_L R}\right) + \frac{1}{D} + \frac{D^2 r_L}{4(M\tau_L)^2 R} + 1}$


Figure 3.24: Open-Loop Gain DCM Approximation Error

resistances. This differs from the ideal and non-ideal curves for the CCM region of operation in that the gain clearly begins to diverge at large output currents. Luckily, the expression for the non-ideal CCM open-loop gain given in (3.44) is not very complex when compared to the non-ideal DCM expression in (3.70). Because of the complexity of the DCM expression, it is very attractive to try and approximate the non-ideal gain to the ideal gain given in (3.57). Figure 3.24 shows the percent error in doing so.

Approximating $G_{d0,nonideal}$ to be $\approx \frac{2V_O}{1+D}$ is perfectly valid for a small load currents. However, the definition for a "small load current" varies on the DC gain of $M = \frac{V_O}{V_S}$: as Figure 3.24 clearly shows, an increasing DC gain causes the error to increase at smaller load current (presented here as larger load resistances). That said, given that operation in DCM is typically dictated by small load currents, this approximation will hold true in most applications; however, it is always wise to verify that the approximation will hold based on the application.

Table 3.2 shows the equations for ω_0 for each transfer function while Figure 3.25 compares them graphically with a varying load resistance. Similar to the open-loop gain comparison plot (Figure 3.23), the ideal and non-ideal DCM ω_0 curves exhibit almost identical behavior for large load resistances. Due to the complexity of the non-ideal expression in (3.73), it would be very beneficial to be able to approximate



 Table 3.2: Comparison of Transfer Function Resonant Frequency Equations

Figure 3.25: Resonant Frequency Comparison for Varying Load Resistances with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and M = 1.6.

 $\omega_{0,nonideal}$ to $\omega_{0,ideal}$. Thus, a graphical comparison is presented in Figure 3.26.

As was observed in the approximation error graph for G_{d0} in Figure 3.24, the approximation holds best for small load currents and small DC conversion ratios. This behavior is also exhibited for the resonant frequency approximation in Figure 3.26.

Next, Table 3.3 compares the expression for the quality factor, Q, for each transfer function while Figure 3.27 graphically compares them for varying load resistances. The Q-factor is an important consideration for the small-signal response since it gives an indication of how stable the system actually is. In canonical second-order systems, the step-response can be plotted just through knowledge of ω_0 and Q alone. When



Figure 3.26: Resonant Frequency DCM Approximation Error

Q > 0.5, the system is said to be *underdamped* in that oscillations will occur when the system is stepped, but they will die out after a certain interval. Conversely, when Q < 0.5, the system is *overdamped* which causes the system to be slow, but no 'ringing' will occur. However, since the boost converter has second order zeros associated with its transfer function, the step-response cannot be extracted as easily (although the underlying canonical second-order effects are still observable). Despite the added complexity due to the presence of $\omega_{z,1}$ and $\omega_{z,2}$, the general trend is still the same for Q in that aiming for a value around 0.5 is most ideal (as it has the fastest rise-time and settling-time). Unfortunately, it's incredibly difficult to observe the effects of various parameters on the value of Q due to the complexity (especially for the non-ideal DCM case in (3.74)).

As with the previous parameters of G_{d0} and ω_0 , the error introduced by approximating $Q_{nonideal}$ to Q_{ideal} is displayed in Figure 3.28. Again, as with the previous parameters, for small load currents the approximation is valid.

Finally, Table 3.4 compares the expression for the zeros of each transfer function while Figure 3.29 graphically compares them for varying load resistances. Figure 3.30 shows the error in using an approximation of $\omega_{z,1,nonideal} \approx \omega_{z,1,ideal}$. As was the case for all previous parameters, the approximation is valid for small load currents.

The result of comparing each transfer function parameter is that for small load cur-

	Q
CCM (Ideal)	$R(1-D)\sqrt{\frac{C}{L}}$
DCM (Ideal)	$\frac{1}{\omega_0} \left[\frac{4(M\tau_L)^2 (1+D)}{4RCDM^2 \tau_L^2 + \frac{L}{R} (D^4 + 4D^2 M \tau_L + 4M^2 \tau_L^2) + \frac{L}{R} D^3} \right]$
CCM (Non-Ideal)	$\sqrt{LC} \frac{\sqrt{(R+r_C)(R(1-D)^2+r_L)}}{Rr_C C(1-D)^2 + Rr_L C + r_C r_L C + L}$
DCM (Non-Ideal)	$\frac{r_L [D^2 (D^2 + 4M\tau_L) + 4(M\tau_L)^2 + D^3] + 4R(M\tau_L)^2 (1+D)}{\omega_0 Q_d}$

 Table 3.3: Comparison of Transfer Function Quality Factor Equations



Figure 3.27: Quality Factor Comparison for Varying Load Resistances with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and M = 1.6.

Table 3.4:	Comparison	of Tra	nsfer Fu	nction Zer	o Frequency	y Eq	uations
------------	------------	--------	----------	------------	-------------	------	---------

	$\omega_{z,1}$	$\omega_{z,2}$
CCM (Ideal)	$\frac{R(1-D)^2}{L}$	∞
DCM (Ideal)	$-\frac{2(M\tau_L)^2}{D^2\frac{L}{R}\left(\frac{D}{2}+\frac{M\tau_L}{D}\right)}$	
CCM (Non-Ideal)	$\frac{R(1-D)^2 - r_L}{L}$	$\frac{1}{r_C C}$
DCM (Non-Ideal)	$-\frac{r_L D^3 + 2r_L D M \tau_L - 4R(M\tau_L)^2}{L D^3 + 2M \tau_L L D}$	$\frac{1}{r_C C}$



Figure 3.28: Quality Factor DCM Approximation Error



Figure 3.29: RHP Zero Comparison for Varying Load Resistances with $L = 18 \,\mu H$, $C = 4.7 \,\mu F$, $F_S = 350 \,kHz$, and M = 1.6.

rents the non-ideal DCM function can be approximated to the ideal transfer function with the only change being $\omega_{z,2} = \frac{1}{r_C C}$ as opposed to ∞ . Since DC-DC converters are typically pushed into DCM operation due to small load currents, this approximation will be valid for a wide range of loads, provided the DC conversion ratio is not very large. The area in which the approximation is not valid will occur near the transition



Figure 3.30: RHP Zero DCM Approximation Error



Figure 3.31: Pole Zero plot for Non-Ideal CCM and DCM Transfer Functions (RHPZs Not Plotted)

between CCM and DCM. This result is somewhat similar to the modes of operation in a MOS transistor in that the weak-inversion and strong-inversion equations are rather simple but are invalid in a certain boundary between weak and strong [22]. Here, the transistor equations become far more complex and the transistor is said to be in moderate inversion. Another similarity between the approximations for MOS equations and boost transfer functions is that a MOSFET is typically used deep in strong-inversion or deep in weak-inversion. DC-DC converters, too, are either operating deep in CCM or deep in DCM since the converter's characteristics undergo significant alterations when transitioning into either operation mode, as previously shown.

The ability to approximate the non-ideal DCM boost converter to that of an ideal one is very powerful, but can be very dangerous if analysis is not done to ensure that the approximation is valid. At low DC conversion ratios, the approximation will be able to hold for a larger range, but even small increases of said conversion ratio can cause massive increases in error which limits the region at which the approximation is valid.

Another important note is the increased controllability by operating a converter in DCM. In Vorpérian's paper on the Average PWM Model in DCM, he mentions that in State-Space Averaging the transfer function for a converter in DCM is found to be a first-order system. However, as was previously found, the system is *clearly* second-order. The reason for the discrepancy, as discussed by Vorpérian, is that the location of the second pole is far from the origin [19].

Figure 3.31 verifies this through observation of the pole/zero locations of both the ideal and non-ideal CCM and DCM transfer functions. Here, the secondary pole in the DCM case is located at a value much greater than that of the first pole. This causes the value for Q to be small, as proven in Figure 3.27, since resonant peaks are a direct effect of conjugate-pair poles. Since the converter in DCM does not have conjugate-pair poles (directly observable in Figure 3.31), it does not have a resonant peak.

The reason this point is important for controllability is that the DCM transfer function *acts* like a first-order system for the given region of interest (frequencies less than that of the switching frequency). This means the phase-margin will be greater than in CCM (since the CCM transfer function clearly exhibits the effects of a secondorder system) which, in turn, means that the Error Amplifier in Figure 3.1 doesn't have to contribute a large phase-shift in order to guarantee stability. This is the primary reason DCM is such an attractive region of operation for boost converters.

Chapter 4

Converter Control

The reason control is important for switched DC-DC converters is to maintain a stable output at various load conditions. Recall from the non-ideal DC equations for CCM and DCM in (2.10) and (2.29), respectively, that each equation has a clear dependence on the size of the load. Thus, in order to maintain a fixed voltage output, the duty cycle *must* be modulated in order to supply said fixed voltage to varying loads.

There are two control methods to perform such an operation: Voltage Mode Control (VMC) and Current Mode Control (CMC).

Current Mode Control

Current Mode Control will not be analyzed in this thesis; however, it is a very popular and well-documented technique for controlling switched power converters [25–29]. The topology is shown in Figure 4.1. The control method employs two feedback loops: one for the output voltage and one for the inductor current. The operation is rather simple in that the inductor current is monitored through the switch and when it reaches a certain peak value, the switch is turned off.

The benefits of this type of control method are speed, large bandwidth, and simple compensation requirements [24]. However, the cost is in complexity (due to the existence of two feedback loops), and limited load regulation.



Figure 4.1: Boost Converter with Current Mode Control (Adapted from [29])



Figure 4.2: Circuit Implementation of a Voltage-Mode Controlled Boost Converter

Voltage Mode Control

Voltage Mode Control can be implemented with the architecture shown in Figure 4.2, and is the architecture that is dealt with throughout this thesis. While the benefits of CMC are certainly tantalizing, the decreased load regulation performance when compared to an equivalent VMC converter is undesirable for applications which need a regulated output for a large range of loads. The downside is that the converter

will be noticeably slower to respond when compared to CMC techniques and that compensation will typically require more components [24].

4.1 Error Amplifiers

The first requirement of an Error Amplifier (in either the VMC or CMC case) is a stable, load-invariant voltage reference that has little to no temperature variation to serve as V_{REF} . These requirements are usually met quite well by implementing a bandgap reference. Since V_{REF} will typically be different than that of V_{OUT} , a resisitive voltage divider is commonly used to divide the desired output voltage down to the value supplied by V_{REF} .

During operation, the error amplifier is attempting to compensate for small variations in the output voltage hence the loop gain must be very large; however, having a large loop gain is not enough. In order to guarantee a stable output (in that the converter does not begin to oscillate), the Barkhausen Stability Criterion states that circuit cannot have any frequency that satisfies both of the following conditions:

- 1. The magnitude of the Loop Gain, $|A\beta|$, is = 1.
- 2. The shift in phase around the loop has a net change of 180° (noting that the inverting terminal of the amplifier contributes to an additional 180° phase shift around the loop).

The most common way to verify this criterion is to observe the Bode plot of the system, which will be done in the next chapter. However, in the context of this chapter, it's important to understand what will cause Barkhausen's Criterion to hold true and, most importantly, how to ensure that it is satisfied with careful error amplifier selection.

4.1.1 Op-Amps vs. OTAs

In most switched converter designs, Op-Amps are employed as the Error Amplifier due to their availability as "off-the-shelf" parts. The different architectures associated with them are very well documented [21,23]; however, while not ill-suited for the job, Op-Amps are certainly unnecessary. Consider what the error amplifier is driving: an input pin of a Comparator. Since a comparator will have high input impedance, it is completely unnecessary to drive it from a low-impedance source (in this case, the Op-Amp).

For integrated circuits, utilizing an Op-Amp instead of an OTA is a very ill-advised design choice. The result is an increase in required circuitry as well as a decrease in response time due to the output voltage buffer. A further benefit to using OTAs is that the gain of the amplifier can be easily set by varying the value of a resistive load. In fact, for control circuitry built on ICs, OTAs are overwhelmingly used over Op-Amps for all of these reasons [3, 4, 6, 28, 29].

Since this focus of this work is on boost converters in an integrated circuit context, the analysis of error amplifier architectures will focus around those utilizing OTAs. Note that for all equations in this chapter,

$$K = g_m R_O \frac{R_B}{R_T + R_B} \tag{4.1}$$

4.1.2 Lag Compensation (Proportional-Integral)

The Lag Compensator is given in Figure 4.3 and has a transfer function given by (4.2). The feature of this type of compensator is that it provides a phase "lag" at a given peak frequency (determined by the geometric mean of the pole and zero frequency,



Figure 4.3: Lag Compensator Circuit

 $\sqrt{f_z f_p}$ [21]). The DC gain of the system is just the gain of the OTA, as shown in (4.2), while the pole and zero locations are given in (4.3) and (4.4), respectively. Note that for (4.2) to be true, $R_O \gg R_Z$.

$$\frac{V_c}{V_o} = K \left[\frac{sR_Z C_Z + 1}{sR_O C_Z + 1} \right]$$
(4.2)

$$|f_p| = \frac{1}{2\pi R_O C_Z} \tag{4.3}$$

$$|f_z| = \frac{1}{2\pi R_Z C_Z} \tag{4.4}$$

The benefit of using a lag compensator, also known as a proportional-integral (PI) controller, is that it can be used to boost the gain of a system without affecting the behavior near the 0 dB crossing point. As such, it's ideal for converters that already have a good phase margin, but just need the DC gain boost in order to eliminate steady-state error.

The primary requirement when designing a lag compensator is to guarantee that the controller's zero frequency is greater than the pole frequency. This is easily done by simply making R_Z much smaller than R_O . As previously mentioned, the frequency at which the controller produces its minimum phase is given by the geometric mean of



Figure 4.4: Bode Plot for Lag Compensator Circuit with $R_O = 6 k\Omega$, $C_Z = 100 nF$, $g_m = -1 S$, $R_T = 2 M\Omega$, $R_B = 500 k\Omega$ and R_Z swept with values of 100Ω , 500Ω , and $3 k\Omega$.

the pole and zero frequency, or $\sqrt{f_z f_p}$. In terms of circuit parameters, this frequency location is given by equation (4.5). Likewise, the total phase shift can be easily calculated using equation (4.6). For example, for a ratio of $\frac{R_Z}{R_O} = \frac{1}{2}$, the phase lags by approximately 20°. A decrease in the ratio by a factor of 2, however, increases the phase lag by 15° (to 35°) - a total change of 75%.

$$f_{min} = \frac{1}{2\pi C_Z \sqrt{R_O R_Z}} \tag{4.5}$$

$$\phi_{min} = \sin^{-1} \left(\frac{R_Z - R_O}{R_Z + R_O} \right) \tag{4.6}$$

4.1.3 Lag Compensation Plus Pole

By adding a capacitor, C_C , in parallel with the series RC string, a high-frequency pole can be added to the system, as shown in Figure 4.6. In terms of equations, this circuit can be shown to have a transfer function equivalent to equation (4.7) where $R_O \gg R_Z$ and $C_Z \gg C_C$. Note that the only difference between this compensator and the one shown previously in Figure 4.4 is the addition of the high frequency pole at $f_{p,2} = \frac{1}{2\pi R_Z C_C}$.

$$\frac{V_c}{V_o} = K \frac{sR_Z C_Z + 1}{s^2 R_O R_Z C_C C_Z + sR_O C_Z + 1}$$
(4.7)

The benefit of this circuit, rather obviously, is that is provides a high-frequency pole which will help attenuate signals near the switching frequency of the circuit. For this reason, it is a very popular choice in switched-converter controller design [3,21,29].



Figure 4.5: Lag Compensator Circuit



Figure 4.6: Bode Plot for Lag Compensator Circuit Plus Pole with $R_O = 6 k\Omega$, $C_Z = 100 nF$, $C_C = 1 nF$, $g_m = -1 S$, $R_T = 2 M\Omega$, $R_B = 500 k\Omega$ and R_Z swept with values of 100Ω , 500Ω , and $1 k\Omega$.

The sizing of C_C will typically be chosen such that the high-frequency pole cancels out the high-frequency zero caused by the ESR of the output capacitance of the converter. This allows for switching frequencies and any sub-harmonic oscillations (a prominent problem in Current-Mode Control [24, 29]) to be attenuated such that they have a negligible effect on converter performance.

4.1.4 Lag Compensation Plus Pole with Feedback

A modification of the lag compensator with the high-frequency pole is shown in Figure 4.7. The modification is simply to take the RC string and connect it between the inverting terminal of the OTA and the output. The benefit is that the capacitor sizes can be reduced which, in an integrated circuit context, will help to save layout area. The downside is that R_Z needs to be made much larger than R_O which will typically require very large values.

The equation for this controller is shown in (4.8) with the relationship to the compensator without feedback given in (4.9) and (4.10) where variables denoted by the subscript fb indicate it is part of the feedback architecture while the subscript nfb is for the architecture without feedback. Note that C_C , R_O , and g_m are all assumed



Figure 4.7: Lag Compensator Circuit with Feedback Zero



Figure 4.8: Bode Plot for Lag Compensator Circuit Plus Pole with $R_O = 6 k\Omega$, $C_Z = 250 fF$, $C_C = 1 nF$, $g_m = -1 S$, $R_T = 2 M\Omega$, $R_B = 500 k\Omega$ and R_Z swept with values of 40.7 $M\Omega$, 218 $M\Omega$, and 480 $M\Omega$.

to not change between the two architectures and are requirements for (4.9) and (4.10) to hold true. A further assumption is that $g_m R_O \gg 1$.

$$\frac{V_c}{V_o} = K \frac{sC_Z R_Z + 1}{s^2 R_O R_Z C_C C_Z + s(R_Z C_Z + R_O C_C - g_m R_O (R_B || R_T) C_Z) + 1}$$
(4.8)

$$R_{Z,fb} = R_{Z,nfb} \left(\frac{g_m R_O(R_B || R_T)}{R_{Z,nfb} - R_O} \right)$$
(4.9)

$$C_{Z,fb} = \frac{R_{Z,nfb}}{R_{Z,fb}} C_{Z,nfb} \tag{4.10}$$

A Bode plot for this circuit is presented in Figure 4.8 and is identical to the plot without feedback in Figure 4.6. The only circuit parameters to change were an increase in R_Z and a decrease in C_Z . The reduction in C_Z is particularly tantalizing from a design standpoint because, as mentioned earlier, it can help to reduce layout size. However, the massive increase in R_Z can be problematic for processes that lack highly resistive layers that can be used for resistors in the Meg-Ohm range. A potential solution to mitigate this problem would be to move R_Z off chip. Since the value will typically be very large, any increased resistance due to external pin contacts will be negligible.

4.1.5 Lead Compensation (Proportional-Derivative)

Lead compensation doesn't have much use in switched converter circuits due to their increase in gain at high-frequencies, but they're worth talking about in order to provide a complete overview of controller architectures.

The circuit used to implement a lead controller is shown in Figure 4.10. The



Figure 4.9: Lead Compensator Circuit



Figure 4.10: Bode Plot for Lead Compensator Circuit with $R_O = 6 k\Omega$, $C_1 = 30 pF$, $C_C = 10 pF$, $g_m = -1 S$, $R_B = 500 k\Omega$, and R_T swept with values of $2 M\Omega$, $8 M\Omega$, and $24 M\Omega$.

transfer function for this circuit is given in (4.11).

$$\frac{V_c}{V_o} = K \left[\frac{sR_T C_1 + 1}{(s(R_B \| R_T)C_1 + 1)(sR_O C_C + 1)} \right]$$
(4.11)

The phase peak occurs at a frequency given by (4.12) while the total phase boost is given in (4.13). The primary benefit to the lead controller is that it can provide a phase boost at a given frequency which can help to either boost the phase margin, or increase the bandwidth of the system. However, the cost is that higher frequencies also see a boost in gain which is undesirable for switched converters since switching noise and subharmonic oscillations can begin to affect the output voltage and have an undesirable effect on EMI [23]. However, if output voltage quality can be neglected in favor of speed, the lead controller is a solid controller choice.

$$f_{max} = \frac{1}{2\pi C_1 \sqrt{(R_B \| R_T) R_T}}$$
(4.12)

$$\phi_{max} = \sin^{-1} \left(\frac{R_T}{R_T + 2R_B} \right) \tag{4.13}$$



Figure 4.11: Lag-Lead Compensator Circuit

4.1.6 Lag-Lead Compensation (Proportional-Integral-Derivative)

The lag-lead compensator, as shown in Figure 4.11, does just as the name suggests: first lags the phase, then leads it. This architecture is simply a merger of both the lag and lead compensators shown in Figures 4.3 and 4.9, respectively. The transfer function for this controller can be found below in (4.14). Note the familiar pole and zero locations from the lag and lead transfer functions in (4.2) and (4.11), respectively. It is assumed that $R_O \gg R_Z$.

$$\frac{V_c}{V_o} = \frac{g_m R_B}{s(R_B + R_T)(C_Z + C_C)} \frac{(sR_T C_1 + 1)(sR_Z C_Z + 1)}{(s\frac{R_Z C_Z C_C}{C_Z + C_C} + 1)(s(R_T \| R_B)C_1 + 1)}$$
(4.14)

The Bode plot for this controller is shown in Figure 4.12. As mentioned previously, the point of this control scheme, as is evident from the graph, is to first lag the phase and then boost it. The benefit to using a lag-lead controller is to obtain the high low-frequency gain that's indicative of a lag controller as well as the phase boost from the lead architecture that allows for an increase in phase margin. The cost, rather obviously, is that it increases the number of components required to implement a controller and, less obviously, is far more sensitive to pole and zero locations.

In order for a lag-lead controller to be properly designed, $|f_{p,1}| < |f_{z,1}| < |f_{z,2}| < |f_{p,2}|$ where the pole and zero relationships are given in (4.15) through (4.19). The



Figure 4.12: Bode Plot for Lag-Lead Compensator Circuit with $C_1 = 0.8 \, pF$, $C_C = 2 \, nF$, $g_m = -10 \, \mu S$, $R_B = 500 \, k\Omega$, $R_T = 2 \, M\Omega$, $C_Z = 300 \, nF$, and R_Z swept with values of $100 \, \Omega$, $500 \, \Omega$, and $1 \, k\Omega$.



Figure 4.13: Pole-Zero Plot for Lag-Lead Compensator Circuit with $C_1 = 0.8 \, pF$, $C_C = 2 \, nF$, $g_m = -10 \, \mu S$, $R_B = 500 \, k\Omega$, $R_T = 2 \, M\Omega$, $C_Z = 300 \, nF$, and R_Z swept with values of $100 \, \Omega$, $500 \, \Omega$, and $1 \, k\Omega$.

third pole, $f_{p,3}$ is less sensitive to its location as it is a much higher frequency than the other values. Typically, this is placed at one half the switching frequency in order to properly attenuate the noise caused by switching [30].

Figure 4.13 shows the pole and zero locations on an imaginary plane. Note the

progression of pole \rightarrow zero \rightarrow zero \rightarrow pole from right to left, which is needed in order for the lag-lead compensator to function correctly. Also, as mentioned previously, there is a high-frequency pole that sits at a location that is far enough away from the other poles and zeros where it will not have a heavy influence on the converter (other than to attenuate events near the switching frequency).

$$f_{p,1} = 0 (4.15)$$

$$f_{p,2} = \frac{1}{2\pi (R_B \| R_T) C_1} \tag{4.16}$$

$$f_{p,3} = \frac{C_C + C_Z}{2\pi R_Z C_Z C_C}$$
(4.17)

$$f_{z,1} = \frac{1}{2\pi R_Z C_Z}$$
(4.18)

$$f_{z,2} = \frac{1}{2\pi R_T C_1} \tag{4.19}$$

4.2 Controller Selection

Being able to determine what controller architecture is best suited for a given switcher design is paramount in guaranteeing a stable and well-performing switched converter. Each controller topology, like any circuit, has a series of advantages and disadvantages associated with them that need to be weighed against device requirements. These advantages and disadvantages have been discussed in each controller architecture description, but to fully understand the implications it's best to observe their use in typical circuits.

As Basso describes in his book, the lag-plus-pole architecture is the most popular as it provides good low-frequency gain as well as the ability to cancel the effect of the zero due to the series resistance on the filter capacitor [21]. For example, consider a converter in DCM with a zero caused by the aforementioned capacitor ESR. By applying a basic lag architecture (with only one pole and zero), the converter has a frequency response shown in Figure 4.14. Here, the ESR of the filter capacitor causes the gain to level off at high frequency resulting in a crossover frequency near $300 \, kHz$ which is near the frequency many switchers are typically operated at. Clearly, this is not ideal as the switching noise will either be amplified by the system (for switching frequencies under $300 \, kHz$) or just slightly attenuated (for frequencies above $300 \, kHz$).

To solve this problem, C_C is added in parallel with R_O on the controller to create the lag-plus-pole compensator. Without modifying any other parameters in either the boost circuit or compensator, C_C can be chosen such that is cancels out the zero caused by the boost ESR. In doing so, the crossover frequency is decreased down to around $60 \, kHz$ with a $-20 \, dB$ attenuation at the previous crossover frequency of $300 \, kHz$ which is a huge improvement for only adding in a $5 \, nF$ capacitor (2000 times smaller than that of C_Z !).

The lag-lead compensation architecture is best suited for converters with conjugatepair poles such as the CCM boost converter. The primary reason for this is that it provides the needed low-frequency gain boost that minimizes steady-state error while also providing the higher frequency phase boost that increases phase-margin. However, designing a controller that guarantees CCM stability is still no trivial task. The resonant frequency peak is still quite a problem as the large phase shift (180° worst-case) will require a large phase boost from the controller in order to achieve any semblance of stability.

Since the controller in a switched converter has well defined architectures, as has



Figure 4.14: Bode Plot for Lag Compensator with DCM Boost with $G_{d0} = 5$, $C_O = 10 \,\mu F$, $R_{ESR} = 0.5 \,\Omega$, Q = 0.05, $\omega_0 = 16 \,kHz$, $R_O = 3 \,k\Omega$, $R_Z = 100 \,\Omega$, $C_Z = 10 \,\mu F$, $g_m = -1S$, $R_T = 2 \,M\Omega$, and $R_B = 500 \,k\Omega$



Figure 4.15: Bode Plot for Lag Compensator Plus Pole with DCM Boost with $G_{d0} = 5$, $C_O = 10 \,\mu F$, $R_{ESR} = 0.5 \,\Omega$, Q = 0.05, $f_0 = 16 \,kHz$, $R_O = 3 \,k\Omega$, $R_Z = 100 \,\Omega$, $C_Z = 10 \,\mu F$, $C_C = 50 \,nF$, $g_m = -1S$, $R_T = 2 \,M\Omega$, and $R_B = 500 \,k\Omega$

been shown throughout this chapter, the only real unknowns are which architecture is best suited for a given converter, as well as what the optimum component values are. As such, it's possible to use AI techniques such as Genetic Algorithms in order to optimize the design of a converter controller. This technique is discussed more in Appendix A.

Choosing a controller topology is highly dependent on converter type as well as operating mode, as shown. Component value selection is extremely critical in order to ensure a robust and repeatable design that is accurate and stable in all operating conditions.

Chapter 5

Full Boost Converter Model Analysis

Utilizing the models derived throughout the previous chapters, a full small-signal model of a boost converter can be constructed. This will allow for a thorough investigation of what parameters have the largest impact on boost converter stability. This, in turn, will allow for more comprehensive studies in ways to test said stability as well as present useful debugging methods.

Throughout this chapter, both the CCM and DCM converters will be analyzed. A baseline design will be presented and various parameters will be modified in order to gauge the sensitivity of frequency response performance metrics to these parameters.

5.1 CCM Converter Model

As mentioned in Chapter 4 and outlined in various sources [15, 21, 23], due to the conjugate-pair-pole intrinsic to CCM Boost Converters the only truly acceptable control topology is that of a PID (*a.k.a.* Lag-Lead) controller. A full circuit implementation is shown in Figure 5.1.

Table 5.1 lists the component values used in this design and the non-ideal component values are as follows: $r_L = 0.8 \Omega$, and $r_C = 0.5 \Omega$.

From the previous analysis in Chapter 4, it's possible to find the pole and zero



Figure 5.1: Boost Converter with PID Control

locations of this converter. The equations for the poles are given in (5.1) [31].

$$s_1, s_2 = \frac{f_0}{4\pi Q} \left(-1 \pm \sqrt{1 - 4Q^2}\right) \tag{5.1}$$

Based on the parameters given in Table 5.1 as well as the parasitic values presented earlier, the values for f_0 , Q, and the pole/zero locations are shown in Table 5.2. The value for Q is larger than 0.5 therefore there is a conjugate pole - as is expected for a CCM converter. The frequency at which the pole exists is given in Table 5.2 as 2.68 kHz. Using (5.1), this occurs at $-\sigma \pm j\omega = -7.2 \times 10^3 \pm j15.2 \times 10^3$ when plotted on the complex plane. The Bode plot and pole-zero plot can be seen in Figures 5.2 and 5.3, respectively.

 Table 5.1: CCM Boost Converter DC Specs

Parameter	V_S	V_O	I_O	L	F_S	C
Value	3V	6V	300 mA	$200 \mu H$	350kHz	$5\mu F$

Table 5.2: CCM Boost Converter Pole and Zero Locations

f_0	Q	$f_{p,1}$	$f_{p,2}$	$f_{z,1}$	$f_{z,2}$
2.68 kHz	1.17	2.68 kHz	2.68kHz	3.34 kHz	63.7 kHz



Figure 5.2: CCM Boost Converter Bode Plot



Figure 5.3: CCM Boost Converter Pole-Zero Plot

In order to properly design this converter and to guarantee stability, the zeros of the PID controller must be placed at or near the resonant frequency, f_0 , which occurs at 2.68 kHz. To do so, the component values included in the equations $\frac{1}{2\pi R_Z C_Z}$ and $\frac{1}{2\pi R_T C_1}$ (from (4.18) and (4.19)) must be chosen such that the ensuing frequency occurs near f_0 .

First, the value for R_T should be chosen. Since the value for this resistor is

constrained by the need for the R_T and R_B string to divide down the output voltage to the reference (which will be 1.3 V used for this example), it's given that $R_T =$ $3.583R_B$. In order to guarantee that the power consumption of the resistive feedback string is minimized, the value for both R_T and R_B must be large. The choice of these parameters is, ultimately, up to the designer so for this controller design the values were selected as $R_B = 560 k\Omega$ which yields $R_T = 2 M\Omega$.

With the selection of R_T , C_1 can be determined to be $C_1 = 30 \, pF$ via (4.19) which was presented again earlier in this section. Next, the values for R_Z and C_Z must be selected. For this design example, it was decided that the second zero frequency would be placed at $3 \, kHz$ which is slightly larger than the resonant frequency of the converter, but still close enough such that dual-zero behavior would still be exhibited near f_0 . By setting $R_Z = 200 \, k\Omega$, the capacitor value becomes $C_Z = 260 \, pF$.

With the values that determine the controller's zero frequency selected, the pole frequencies must be calculated and shown to be valid (recall from Chapter 4 that $f_{z1,2}$ must be less than $f_{p2,3}$). Since $f_{p,2} = \frac{R_T + R_B}{2*\pi R_T R_B C_1}$, it's clear that this pole will be located at $12 \, kHz$. The last pole, $f_{p,3}$ has some flexibility as to where it is placed. Because there is a high-frequency zero from the converter due to the ESR of the filter capacitor, it's wise to place $f_{p,3}$ such that it cancels this zero out. Since $f_{p,3} = \frac{C_C + C_Z}{2\pi R_Z C_C C_Z}$, $C_C = 13 \, pF$.

Finally, the only parameter left is the transconductance of the amplifier itself, g_m . This value can be modified by changing the bias current of the OTA which allows for the gain to be modified in order to achieve both a good phase margin and DC gain (which eliminates steady-state error). In this example, $g_m = 1.03 \mu S$ as it allows for a decent phase margin of $\phi_M = 45^\circ$ and DC gain of 40 dB. The Bode plot, root-locus plot, and step-response are showed in Figures 5.4, 5.5, and 5.6, respectively.

As Figure 5.4 shows, just a slight increase in gain will result in instability due to the large phase change near the converter's resonant frequency. In fact, the gain margin of the circuit is near 2 dB which is *much* smaller than what a commercial design would require. Observation of the root-locus in Figure 5.5 confirms this as it shows that by increasing the gain, the conjugate poles will transition to the RHP which is a characteristic of system instability. However, by decreasing the gain any further, the steady-state error will begin to noticeably worsen. Figure 5.6 shows the problems with low DC-gain: the converter struggles to reach steady-state.

5.1.1 Component Sensitivity

Observing the sensitivity of stability to various parameters is important in order to understand what types of component tolerances are acceptable in a switched converter.



Figure 5.4: CCM Boost Converter Bode Plot with PID Control; $g_m = 1.03 \,\mu S$, $R_B = 560 \,k\Omega$, $R_T = 2 \,M\Omega$, $R_Z = 200 \,k\Omega$, $C_C = 13 \,pF$, $C_Z = 260 \,pF$, $C_1 = 30 \,pF$.



Figure 5.5: CCM Boost Converter Root-Locus with PID Control



Figure 5.6: CCM Boost Converter Step-Response with PID Control

Unfortunately, due to the complexity of the system, it's rather tedious to mathematically derive general sensitivity equations. Even if said expressions were derived, the relationship between the parameter values and stability parameters, such as phase margin, would be difficult to ascertain. As such, a better approach is to observe how these values change graphically.

To begin, the converter design from the previous section will be used again here. First, the tolerances of various components will be defined. The best, nominal, and worst-case values will then be used and plugged into back into the design and the phase margin, gain margin, DC gain, and bandwidth recorded. The values that will be modified are shown in Table 5.3.

Figure 5.7 shows how the phase margin, ϕ_M , changes with different parameter combinations. Note that each division shown on the graph contains two further unmarked corners from the value for L and g_m . As the bars progress from left to right in each load current division, L varies from minimum to nominal to maximum while g_m varies through its values each time per L value.

 Table 5.3:
 CCM Boost Converter Parameter Corners

Component	V_S	I_O	L	g_m
Tolerance	$\pm 40\%$	$\pm 50\%$	$\pm 15\%$	$\pm 10\%$



Figure 5.7: CCM Boost Converter Phase Margin Over Parameter Corners. $V_S = [1.8V, 3V, 4.2V], I_O = [150 \, mA, 300 \, mA, 450 \, mA], L = [170 \, \mu H, 200 \, \mu H, 230 \, \mu H], g_m = [0.93 \, \mu S, 1.03 \, \mu S, 1.13 \, \mu S].$

From this graph, three general trends are readily apparent:

- 1. As the DC conversion ratio increases, the phase margin increases.
- 2. As the load current increases, the phase margin increases.
- 3. As the inductance increases, the phase margin decreases.

The most stable operation region is when $V_S = 4.2 V$ (which translates to M = 1.4) and $I_O = 450 \, mA$. Here, ϕ_M hovers around the 80° mark regardless of the values for either g_m or L. However, this comes at a cost. Figure 5.8 depicts the bandwidth of the converter across the same corners and, as can be seen, bandwidth decreases with increasing phase margin. This is downside of requiring a fast converter: less stability.

Figures 5.9 and 5.10 show the gain margin and the DC Gain of the converter, respectively. The gain margin has a slightly different parametric relationship with the load current when compared to the phase margin. Here, larger load currents cause smaller values of gain margin. This is due to the fact that the phase margin is only increasing because the zero-crossing frequency is being moved to a lower value (i.e., lowering the bandwidth). The phase at these lower frequency values is located



Figure 5.8: CCM Boost Converter Bandwidth Over Parameter Corners. $V_S = [1.8 V, 3V, 4.2 V], I_O = [150 mA, 300 mA, 450 mA], L = [170 \mu H, 200 \mu H, 230 \mu H], g_m = [0.93 \mu S, 1.03 \mu S, 1.13 \mu S].$



Figure 5.9: CCM Boost Converter Gain Margin Over Parameter Corners. $V_S = [1.8V, 3V, 4.2V], I_O = [150 \, mA, 300 \, mA, 450 \, mA], L = [170 \, \mu H, 200 \, \mu H, 230 \, \mu H], g_m = [0.93 \, \mu S, 1.03 \, \mu S, 1.13 \, \mu S].$

just before the 'drop-off' caused by the conjugate-pair pole and as can be observed in the Bode plot back in Figure 5.4. Because there is a rather rapid decrease in phase



Figure 5.10: CCM Boost Converter DC Gain Over Parameter Corners. $V_S = [1.8 V, 3V, 4.2 V], I_O = [150 mA, 300 mA, 450 mA], L = [170 \mu H, 200 \mu H, 230 \mu H], g_m = [0.93 \mu S, 1.03 \mu S, 1.13 \mu S].$

shortly after the zero-crossing, it's conceivable that the phase will equal 0° before the gain drops a sufficient amount to provide a large gain margin. The DC gain, in Figure 5.10, seems to be rather resilient to changes in parameter barring large load currents at large conversion ratios. This is most easily explained by observing the gain of just the switched converter block itself and ignoring the controller. The expression was defined previously in Chapter 3 in equation (3.44). The expression is a function of both the duty cycle and the load resistance. By plugging in conversion ratio into D and the load current into R, (3.44) can be modified as follows in (5.2),

$$G_{d0,CCM} = M V_O \frac{V_O - r_L I_O M^2}{V_O + r_L I_O M^2}$$
(5.2)

It's clear that as both I_O and M increase, the gain of the converter block must decrease since the denominator will become larger than the numerator. Likewise, the larger M is, the smaller (5.2) will become which is why the effect is more prevalent at the larger conversion ratio in Figure 5.10.

5.2 DCM Converter Model

The DCM Converter is shown in Figure 5.11 with a Lag-Plus-Pole Feedback OTA (also considered a PI Controller). The specifications for the converter are shown below in Table 5.4.

As shown in Chapter 3, the DCM boost converter does not have a conjugate pole and thus is easier to control. Table 5.5 shows the locations of the poles and zeros for the DCM converter. Again, because Q < 0.5, the converter does not have any poles with imaginary components.

Figure 5.12 shows the Bode plot for the boost converter while Figure 5.13 shows

 Table 5.4:
 DCM Boost Converter DC Specs

Parameter	V_S	Vo	I_O	L	F_S	C
Value	3V	6V	30 mA	$20\mu H$	350kHz	$5\mu F$

Table 5.5: DCM Boost Converter Pole and Zero Locations

f_0	Q	$f_{p,1}$	$f_{p,2}$	$f_{z,1}$	$f_{z,2}$
9.33kHz	0.06	586 Hz	150kHz	298 kHz	63.7kHz



Figure 5.11: Boost Converter with PI Control



Figure 5.12: DCM Boost Converter Bode Plot



Figure 5.13: DCM Boost Converter Pole-Zero Plot (RHPZ Not Shown)

the pole and zero locations on the complex plane. The convenient reality of the DCM boost converter is that it behaves like a single-pole system at frequencies less than half of the switching frequency. Ideally, a controller would be designed that just raises the DC gain of the system while not disturbing the crossover frequency.

To achieve such a design, the first pole of the controller should be placed at a low frequency while the second pole should be placed such that it minimizes the effect of the converter zero due to the ESR. The zero of the controller should be placed such that it cancels the pole from the converter so that the phase lag it produces can be eliminated. The overall effect is that the controller frequency response becomes more dominant than that of the converter frequency response. This indicates that time constant $R_Z C_Z$ should be equal to $272 \,\mu s$. Choosing $C_Z = 100 \, pF$ results in $R_Z = 1.7 \, M\Omega$.

Choosing the pole locations, however, is not as simple. The complexity of the transfer function for this type of controller, as given in (4.8) in Chapter 4 whose denominator is reproduced below in (5.3), makes it more difficult to ascertain the relationship between parameter values and pole locations. The easiest approach is to plug in the known values of R_Z and C_Z as well as expected values for R_O , g_m , R_T , and R_B . As was the case with the CCM converter, R_T is constrained to be $3.583R_B$ so by setting $R_B = 600 \ k\Omega$, it follows that $R_T = 2.15 \ M\Omega$. Likewise, since R_O must be large and $g_m R_O \gg 1$, good estimates as to what those values will be can be discovered. These parameters largely depend on the architecture of the OTA, but $R_O = 1 \ M\Omega$ and $g_m = 10 \ mS$ are not unreasonable.

$$0 = s^2 R_O R_Z C_C C_Z + s (R_Z C_Z + R_O C_C - g_m R_O (R_B || R_T) C_Z) + 1$$
(5.3)

By taking those calculated values, the expression in (5.3) reduces to:

$$0 = s^2(270C_C) + s(10^6C_C + 0.4) + 1$$

By choosing a secondary pole frequency of 6 kHz, the value for C_C emerges when the magnitude of the above expression is solved for 0 (plugging $j\omega$ in for s). This yields $C_C = 44 nF$. The choice of the secondary pole location is somewhat arbitrary as it just needs to help attenuate high-frequency signals. Here it was also used to move the crossover frequency to a slightly lower value in order to achieve a better phase margin.

Figure 5.14 shows the Bode plot of the whole system. The crossover frequency occurs near the secondary pole of the PI Controller at 6 kHz with a phase margin equal to $\phi_M = 60^\circ$. The root locus of the system is shown in Figure 5.15 and never crosses the imaginary axis which indicates that the system is inherently stable.

However, the poles of the controller can end up 'colliding' on the real-axis causing them to breakaway and become a conjugate pair pole. This is easily controllable by not allowing the gain of the controller to exceed a certain value (most OTAs will be naturally constricted to some operating range anyway). The step response in Figure 5.16 verifies the stability of the system by showing very little oscillation (much different than the CCM step response in Figure 5.6), as would be expected from a



Figure 5.14: DCM Boost Converter with PI Control Bode Plot



Figure 5.15: DCM Boost Converter with PI Control Root Locus



Figure 5.16: DCM Boost Converter with PI Control Step Response

system with a phase margin of 60° .

5.2.1 Component Sensitivity

As was done with the CCM with PID Control Converter, the sensitivity of various frequency-domain measurements will be graphically observed in relation to varying parameters. Table 5.6 shows the tolerances used for various components.

Figure 5.17 shows how the phase margin changes when each component is varied. Unlike in the CCM case, it's clear that as the load current increases, the phase margin decreases. However, the change is very slight as it takes an 80% increase in the load current to drop the phase margin by only 15°. Likewise, this value is very resilient to changes in conversion ratio and inductor size as the only major value shift is due to the load current.

Since the phase margin decreases with increasing load current, it follows that the bandwidth is likely increasing which is verified in Figure 5.18. Again, as was the case with the phase margin, the bandwidth is relatively immune to changing any other

 $\begin{array}{|c|c|c|c|c|}\hline \textbf{Component} & V_S & I_O & L & g_m \\\hline \textbf{Tolerance} & \pm 40\% & \pm 80\% & \pm 25\% & \pm 20\% \\\hline \end{array}$

 Table 5.6:
 DCM Boost Converter Parameter Corners


Figure 5.17: DCM Boost Converter Phase Margin Over Parameter Corners. $V_S = [1.8 V, 3V, 4.2 V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H, 25 \mu H], g_m = [8 mS, 10 mS, 12 mS].$



Bandwidth for Corners of DCM Boost

Figure 5.18: DCM Boost Converter Bandwidth Over Parameter Corners. $V_S = [1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H, 25 \mu H], g_m = [8 mS, 10 mS, 12 mS].$

parameter outside of the load current which is a testament to the robustness of the VMC-DCM Converter.



Figure 5.19: DCM Boost Converter Gain Margin Over Parameter Corners. $V_S = [1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H, 25 \mu H], g_m = [8 mS, 10 mS, 12 mS].$



Figure 5.20: DCM Boost Converter DC Gain Over Parameter Corners. $V_S = [1.8V, 3V, 4.2V], I_O = [6 mA, 30 mA, 54 mA], L = [15 \mu H, 20 \mu H, 25 \mu H], g_m = [8 mS, 10 mS, 12 mS].$

The gain margin in Figure 5.19 exhibits a slightly different trend than seen with the phase margin. Here, this value decreases with an increasing conversion ratio. At the larger conversion ratio of M = 3.3, the gain margin decreases with increasing

load current; however, for all other conversion ratios the gain margin stays relatively constant. This is possible since the slight increase in phase margin indicates that the zero frequency crossing occurs at a point with a small phase slope. It is likely, therefore, that the zero crossing also occurs just before a rapid decrease in phase (which would be caused by the high-frequency pole in this design). Since the phase margin is hovering around a value of 50° , the phase just needs to shift by that much to hit the 0° phase point. Since the gain plot will not exhibit a decrease in gain due to a pole until the phase has shifted by 45° , this causes a low value for gain margin. That said, the minimum gain margin shown for any corner in the DCM case is roughly *twenty times* larger than the *largest* gain margin in the CCM case!

Finally, the DC Gain plot in Figure 5.20 shows an almost complete invariance to component values and has a value in all instances around $55 \, dB$. This shows the versatility of the DCM converter in that it can achieve the same level of steady-state error rejection over a wide range of operating regions.

5.3 Performance Comparisons

The difficulty of controlling a CCM converter presents many stability issues, as shown throughout this chapter. It takes a lot of design effort to create a controller that can achieve an acceptable amount of steady-state error rejection (DC Gain), speed (Bandwidth), and stability (Phase/Gain Margin). Even with a very careful design and very high tolerance controller components (which is unlikely in an integrated circuit), the external component variations can have a detrimental impact on converter performance which limits the effective operation range.

The DCM converter, in contrast, exhibits significantly less performance degradation with parameter variation. In addition, the design of the controller itself requires fewer components and is less susceptible to cause instability due to the simple single-pole nature of the DCM converter itself. This makes VMC Boost Converters significantly more desirable for mobile integrated circuit applications where external components can have large tolerances and where the operating range must be rather wide in order to function correctly in different conditions.

Chapter 6

Stability Measurement Techniques

The ability to accurately measure the stability of a DC-DC converter is vital in not only correlating hardware with simulation, but also in gauging the viability of design inclusion in a commercial product. Unfortunately, traditional analytic methods of measuring loop stability involve physically 'breaking' the loop and applying a test signal to one end and calculating the return signal. In a switched DC-DC converter, this is simple not possibly due to the large loop gain that serves to correct any variations from the steady-state value. By breaking the loop, the steady-state operating point will become disturbed and the very small difference in operating points will cause the high-gain controller to ramp up and saturate at its supply voltage, rendering the converter useless.

In 1975, Middlebrook proposed a method to deal with this problem [32]. Instead of breaking the loop to apply the test signal, the signal would be injected at a point where it would be functionally equivalent to breaking the loop. This will be described more thoroughly in Section 6.2.

Middlebrook's method for determining loop stability is very attractive, but it is limited by the need for certain impedance assumptions to hold true. To circumvent this, simple transient responses can be measured and the stability can be ascertained from the step and/or impulse responses [33]. This transient behavior can also be used to obtain frequency response characteristics by means of cross-correlation and system identification [34,35].

6.1 Step Response

Transient step responses are very important waveforms to observe when trying to determine the stability of any closed-loop system, let alone a switched converter. The reason is because unlike traditional AC-Analysis methods, it does not rely on circuits operating in some linear region: it just shows you *exactly* how the circuit will behave given some stimulus. As such, it's an important type of measurement that should be discussed before analyzing any other methods.

Switched DC-DC Converters are usually designed to operate within fairly large regions. This may include a variety of input voltages, output voltages, load currents, or any combination of all three. As such, there can be *many* different combinations that could cause instability.

One of the simplest ways of testing supply stability is by stepping the load [33]. This is done by quickly changing the load current so as to put strain on the converter and force it to rise back up to a stable state. This can be done quite easily with modern test equipment, such as a Keithley 2400 SourceMeter SMU. By stepping the load from some point near the minimum load operating range to the maximum load current, a step response can be recorded on an oscilloscope and captured for analysis.

The type of step response that the converter exhibits offers some insight as to its



Figure 6.1: DCM Boost Converter Load Step from $10 \, mA$ to $20 \, mA$ with an Underdamped Step Response

stability. Figure 6.1 shows the response for a Boost Converter in what is known as an underdamped condition (the thick waveform is due to the high-frequency-switching). Intuitively, the more 'ringing' the waveform exhibits, the less stable the converter. Figure 6.2 shows an overdamped case where there is no ringing and the converter takes a longer time to increase to its steady-state voltage

Unfortunately, the amount of ringing that the converter exhibits is just about all the information that can be garnered from a step response. While it offers insight into the stability, it provides no insight as to what the phase margin may be nor what the frequency response actually looks like. This requires the designer to manually observe the response and deem it either acceptable or unacceptable in terms of stability which is very difficult, if not impossible, for products meant for mass production. As another downside to using this method, it is possible that the ringing exhibited by the controller can be 'buried' in the switching noise from the converter. As such, at a large voltage scale (on an oscilloscope), it's possible that the converter response *appears* stable when, in fact, there could be a significant amount of ringing.



Figure 6.2: DCM Boost Converter Load Step from $10 \, mA$ to $20 \, mA$ with an Overdamped Step Response

6.2 Frequency Response

The frequency response, unlike the step response, provides much more information about the circuit under test. Characteristics such as resonant frequencies, phase margin, and gain margin can all be extracted from a frequency response which allows a quantitative measure of the stability of the system as opposed to simply comparing waveform shapes or observing the number of peaks that a step or impulse response has.

Unfortunately, the ability to accurately measure the loop gain and phase of a switched converter is dependent on a few criteria:

- The feedback loop is not broken.
- Measurements are made with high-accuracy equipment.
- Injected signals do not alter the steady-state operation of the converter.

6.2.1 Injecting a Voltage

Injecting a voltage signal without disturbing the steady-state characteristics of the converter was first described in Middlebrook's paper in 1975 [32]. In normal AC circuit analysis on a closed-loop system, a point within the loop is broken in order to apply a test signal. The transfer function of the circuit is then characterized by the ratio of the return signal, V_r , to injected signal, V_i . Since these two potentials exist on either end of the point at which the loop was broken, as shown in Figure 6.3, it can be said that $T(s) = \frac{V_r}{V_i}$.

However, as previously mentioned, breaking the loop within a switching converter (or any system with large loop gain) is not possible without disturbing the steadystate operating point. To alleviate this problem, Middlebrook proposed injecting the voltage in series with the loop, as shown in Figure 6.4. Since this voltage will still have a potential at the injection node of V_i and a return potential of V_r , this method is functionally equivalent to physically breaking the loop while keeping the steady-state operation undisturbed (besides the injected signal, of course).

Unfortunately, there is a *caveat* to this approach: the equivalent impedance at the positive injection node must be greater than the equivalent impedance at the



Figure 6.3: Voltage Injection via Opening the Loop (Adapted from [32]).



Figure 6.4: Voltage Injection via Middlebrook's Method.

negative injection node. This means that the return signal should be located at a node that approximates to an ideal voltage source. Fortunately, such a point exists on a switched converter between the output of the power conversion stage and the input of the resistive feedback string. Being a voltage source, the output impedance of the conversion stage is very low while the resistive feedback string is, typically, on the order of Meg-Ohms of resistance. As such, it is the ideal place for voltage injection via Middlebrook's method.

6.2.2 Injecting a Current

In the same paper, Middlebrook also proposed current injection as opposed to voltage injection. Similarly to the voltage injection method, the transfer function is characterized by the ratio of the return signal, I_r , to the injected signal, I_i . Since this causes a third current I_x , it's possible to simply apply a known current as the test signal at I_x and measure the current at the injection and return branches to extract the transfer function. Again, however, there is a *caveat*: the impedance at the injection branch must be much less than the impedance at the return branch. The reason for



Figure 6.5: Current Injection via Opening the Loop (Adapted from [32]).



Figure 6.6: Current Injection via Middlebrook's Method.

this, much like the reasoning for the voltage injection, is that the return branch must approximate to a current provided by an ideal current source. As such, an adequate injection point would be at the output node of an OTA which has very large output impedance as it is, essentially, a voltage-controlled current source.

6.2.3 Response Measurement

The easiest way to measure the frequency response via Middlebrook's Method is with a Frequency Response Analyzer (FRA) [44]. These analyzers provide an oscillator output that can sweep through frequency ranges and can make very precise voltage measurements at the frequency being swept which allows for very accurate magnitude and phase plots of the converter under test.

In order to apply the voltage injection, the source should be isolated from the circuit through an isolation transformer. Since the positive and negative leads of the source cannot be placed at the same node, it must be applied across some injection resistor that is placed in series with the loop. This can be done in the resistive feedback string of the switched converter with no penalty on converter performance



Figure 6.7: Voltage Injection Test Setup.

since the injection resistor will be in the tens-to-hundreds of Ohms range while the feedback string value is orders of magnitude greater. Figure 6.7 shows the required test set-up for frequency response analysis of a switched converter.

6.3 Frequency Response via Cross-Correlation

The idea of using Cross-Correlation as a way to extract the frequency response of switched DC-DC converters is the subject of much research [34–41]. The idea is attractive as it provides the ability to have a BIST circuit on chip which would allow for automated frequency response testing. While the design and implementation of such a circuit fall beyond the scope of this thesis, the underlying principles will be described here.

6.3.1 Cross-Correlation

Cross-Correlation is a statistical process relating the correlation between two signals. Autocorrelation is simply the cross-correlation of a signal with itself. For continuous systems, the cross-correlation of signals x(t) and y(t) are given as:

$$R_{xy}(t) = \int_0^\infty x^*(\tau) y(t+\tau) \ d\tau$$
 (6.1)

However, it is highly likely that the signals will be quantized before any crosscorrelation operation occurs so (6.1) must be modified in order to hold true for discrete sets of data. Such an equation is given in (6.2).

$$R_{xy}[m] = \sum_{n=1}^{\infty} x[n]y[n+m]$$
(6.2)

In order to use the cross-correlation method of determining a system's frequency response, an assumption must be made that the converter is an LTI system for small output disturbances. With this assumption, the sampled converter can be described by (6.3) where h[k] is the converter's sampled impulse response, x[k] is the control signal, and v[k] is representative of any disturbances in the system, as reported in [34,36])

$$y[n] = \sum_{k=1}^{\infty} h[k]x[n-k] + v[n]$$
(6.3)

Using (6.3) and plugging it in for y[n+m] in (6.2) yields the cross-correlation of the control and output signals:

$$R_{xy}[m] = \sum_{n=1}^{\infty} h[n] R_{xx}[m-n] + R_{xv}[m]$$
(6.4)

If the assumption is made that the input control signal, x[k] is white noise, the autocorrelation of this signal is given by an ideal delta function, $\delta[m]$ and the crosscorrelation of this noise with external disturbances must be zero since it implies that x[k] and v[k] are statistically independent. Thus, using white noise as x[k], the cross-correlation of x[k] and the converter output, y[k], is simply the discrete impulse response of the system:

$$R_{xy}[m] = h[m] \tag{6.5}$$

By taking the DFT of this cross-correlation, the transfer function of the system can be extracted which allows for the loop gain and phase to be plotted.

6.3.2 Pseudo-Random Binary Sequences (PRBS)

A popular way to generate noise, especially in digitally controlled converters, is through the use of PRBS. These binary sequences have a length as a function of how many bits they contain. For an *n*-bit converter, one period of the PRBS is given by (6.6) where f_k is the clock frequency of the PRBS circuit which is typically set equal to the switching frequency of the DC-DC converter [36].

$$T_{PRBS} = \frac{2^n - 1}{f_k}$$
(6.6)

Figure 6.8 shows an example implementation of an 8-bit PRBS generation circuit and Figure 6.9 shows the output waveform after start up for a small sampling of the PRBS period.

As mentioned previously, white noise has an autocorrelation equivalent to that of an ideal delta function. Therefore, in order for a PRBS implementation to be



Figure 6.8: 8-bit PRBS Circuit.



Figure 6.9: 8-bit PRBS Signal.



Figure 6.10: Autocorrelation of (a) Single-Period 8-bit PRBS, and (b) Four-Period 8-bit PRBS.

appropriate it must also have an autocorrelation that is approximately equal to a delta function. This is proven in Figure 6.10a. Of course, the autocorrelation of the PRBS can not be exactly equal to an ideal delta function since it does have some periodicity, but it is a good approximation. The correlation 'noise' around the approximate delta function in Figure 6.10a causes frequency response measurement errors which negatively impact the accuracy of this type of modeling.

As a solution to this problem Botao Miao *et al.* proposed sampling the output of the converter for a duration of N-PRBS periods and averaging the cross-correlations in order to obtain a more accurate frequency response measurement [34]. The reason this implementation is preferred is most easily explained by viewing Figure 6.10b which shows the autocorrelation of four periods of an 8-bit PRBS. The amplitude is noticeably larger than that of the single-period PRBS autocorrelation which provides a better approximation to an ideal delta function which has an infinite amplitude. In addition, the correlation 'noise' on each side of the approximate delta functions is much smaller which, again, provides a better approximation to ideal delta functions which have zero-amplitude at all locations besides the location of the impulse itself.

6.3.3 Implementation of a PRBS Cross-Correlation Method

In order to properly implement a stability test method that utilizes a cross-correlation and PRBS, the PRBS circuit itself must be injected at a proper point. Since the goal is to use the PRBS to modulate the duty cycle of the converter, the most logical place



Figure 6.11: Injection Point for PRBS in a Switched DC-DC Converter

to inject the signal is into the input of the PWM Generator. This is done as per the circuit in Figure 6.11

The digital PRBS can be superimposed on the analog V_c signal since all the PRBS will do is slightly modify the error voltage at the input of the comparator. As far as the converter itself is concerned, this just appears to be noise that is modulating the duty cycle. The only requirement is that the amplitude of the PRBS signal be constrained such that it does not alter the steady-state operating point of the converter.

Once the PRBS signal is injected into feedback loop of the converter, providing the ability to measure this signal is the required next step. If post-processing of data is possible in a given test environment, the simplest approach would be to place a pin at the output of the PRBS circuit as well as the output of the converter itself and externally sample the signals.

If externally sampling the signals is a non-ideal approach given a particular test environment, another option would be to stream the data to internal memory and use dedicated circuitry to cross-correlate and possibly perform the required DFT as well. From here, the information can be streamed to external measurement equipment. This sort of implementation is rather costly in terms of required design effort as well as in physical chip area. Designs that utilize this type of method are typically done with FPGAs [34, 35, 37, 38] which are, rather obviously, impractical for mobile integrated circuit applications.

6.4 Stability Measurement Comparisons

The method of testing stability by way of cross-correlation is quite appealing due to the ability to have on-chip BIST capabilities. The task of any external equipment would be to simply sample the PRBS and output signal and perform the crosscorrelation and DFT. This form of measurement is very methodical and can easily be automated which helps eliminate user error.

A large negative, however, is that the data must only be correlated across a known number of PRBS periods. In addition, if the converter does not settle within one PRBS period (which is a distinct possibility for low-bit PRBS circuit), the ensuing frequency response data will be inaccurate.

Middlebrook's method, however, simply measures the converter 'as-is'. It does not rely on correlation functions nor any form of transform; it simply sweeps the frequency of an injected voltage, measures the return signal and plots the results. The obvious downside is the lack of automation available since, ultimately, the waveform must be manually captured and stored.

As mentioned earlier, the manual capture of data could have a negative impact on testing time if the frequency response data is required for a product in mass production. The benefit over cross-correlation, however, is the ease of implementation and accuracy of the results.

Chapter 7

Conclusion

Boost converters operated in VMC-CCM benefit from enhanced efficiency and are able to handle larger loads, but suffer from a control standpoint due to the presence of a conjugate pole in its small-signal transfer function. Boost converters operating in VMC-DCM, on the otherhand, have a much simpler transfer function and are much easier to control as a result.

An implication of the control issues associated with VMC-CCM boost converters is that in order to achieve an acceptable DC-gain and phase margin, the crossover frequency must be small, which results in a slow converter. This can be problematic for applications that require high response speeds as the converter may never reach a steady state before critical events occur which can result in circuit malfunction. Thus, VMC-DCM boost converters become more palatable as a power conversion architecture despite the limited load they can handle. By utilizing such an architecture, the required control circuitry can be minimized and the converter as a whole can achieve high DC-gains and large phase margins with a minimal impact on speed.

Ensuring the stability of the converter is important, and observation of the frequency response is critical if stability in all operating regions must be guaranteed. Observation of the step response provides some insight as to whether or not the converter is stable, but it does not offer any quantifiable frequency-response data that can be compared and validated against simulations. Measurement of loop gain and phase can easily be done with external equipment, or with internal circuitry by cross-correlating the output signal with injected white noise.

7.1 Future Work

VMC-CCM boost converters have very non-ideal frequency response characteristics; however, CMC-CCM boost converters behave similarly to VMC-DCM in that no conjugate pole exists. As such, it may be possible to use a combined VMC/CMC approach in order to design a boost converter that transitions seamlessly between CCM and DCM with no performance degradation.

Cross-correlation by means of PRBS injection is a very promising stability analysis technique that could be implemented in a BIST circuit for an integrated circuit. The design and implementation in silicon would help to verify the viability of such a technique as well as help to characterize some of its limitations. Appendices

Appendix A

Use of Optimization Algorithms in Controller Design

Switched DC-DC converters have many frequency response characteristics that must be optimized in order to produce the best product and eliminate stability concerns. As shown throughout Chapters 4 and 5, the design of the error amplifier is critical as it determines the steady-state accuracy, response time, and stability margins (both gain and phase). Unfortunately, amplifier design is not trivial as improving one performance metric will, typically, decrease another. For example, in order to increase the phase margin of a system, one must typically lower the gain in order to compensate. While this would improve stability, it would also decrease steady-state accuracy.

Due to this complexity, switched converter controller design is well suited to solving via optimization algorithms. Electronics design via optimization algorithms is a well-established field of research and shows promise in providing optimal solutions for high-complexity design [49, 50, 53, 55, 57–62]. Two algorithms explored here are Genetic Algorithms (GAs) and Particle Swarm Optimization (PSO).

Both algorithms attempted to optimize the transfer function of a controller by iterating over the parameters R_O , R_T , the number of poles and zeros, and the coefficients of the transfer function. Both algorithms operated on the results of the controller transfer function multiplied by the DCM Boost Converter transfer function (presented previously in Chapter 3).

A.1 Genetic Algorithms

The basic flow of a GA is to initialize a population with random values and evaluate each member of the population against some fitness criteria. Once this is done, the best performing members then 'mate' wherein random 'genes' are selected from each participant and crossed over to produce children that populate the next generation. During this crossover period, there is a small chance of gene mutation which helps to add variability into the gene pool which serves to eliminate localized convergence problems.

There are many different GA implementations, but the one analyzed here is known as the 'Queen-Bee' method [52, 59]. Figure A.1 shows the flow of a Queen-Bee-type GA (abbreviated as QBGA).



Figure A.1: Block Diagram for a Queen-Bee Genetic Algorithm for Optimization of Switched-Converter Controller Transfer Function.

A.1.1 Queen-Bee Genetic Algorithm (QBGA)

The QBGA is an algorithm based around the interactions of bees in a hive. Essentially, there exists a single queen with which all other bees, known as drones, mate. Occasionally a female bee is produced that ousts the current queen and becomes the new queen.

To implement this in software, drones are created with completely randomized genes. As an initialization stage, each drone competes with each-other in a tournamentstyle fashion and whichever is the best is selected as the queen.

Next, all of the drones mate with the queen and produce two offspring. To do so, a set of genes are randomly selected from each bee's chromosome and crossed over. If the selected genes are grouped into a category called 'A' and the ones not chosen are in 'B', one offspring will receive the 'A' group from the queen and the other will receive it from the drone. The remaining genes are taken from group 'B' of the *opposite* parent. These two offspring then compete with each other and the best survives.

Once each drone mates with the queen, there should be N offspring for N drones. The offspring then competes with each other and the best is selected as a 'virgin queen'. This virgin queen then competes with the current queen and whoever is the best proceeds on to the next generation as the nest queen. This process then repeats with randomized drones.

However, during gene crossover, it is not sufficient to *only* crossover the genes as there must be some probability of gene mutation in order to guarantee variability in the gene pool to avoid converging on local optima. To do so, most algorithms implement a constant that determines the gene mutation probability [52, 53, 59].

In order to speed up convergence, a variable mutation probability was used in this implementation based on the age of the nest queen as an alternative to simply choosing a constant. As the nest queen moves from generation to generation, it is likely that the QBGA has either converged on the best solution or it could be stuck on a local optima. By setting the mutation probability as a variable based on how many generations the nest queen has existed for, more and more variability will be introduced into the gene pool. The goal here is to attempt to push the algorithm out of any potential locally optimal points. If the program has, in fact, found a global



Figure A.2: Comparison of QBGA Convergence with a Varied and Constant Rate of Mutation.

optimum, then an increased mutation rate should have no effect on the solution. The pseudo-code for such a mutation algorithm is described below:

Algorithm A.1: Variable QBGA Mutation Based on Queen Age	<u>)</u>
for gene in crossover do	
$ \begin{array}{c c} \mathbf{if} \ rand() \geq C^* queenAge \ \mathbf{then} \\ \ gene = mutate(gene); \end{array} \end{array} $	
end	
crossover[i] = gene;	

 \mathbf{end}

By setting C to some fractional value, it will eventually converge to have a 100% mutation rate when $\frac{1}{C} = queenAge$. However, this value *must not* be set too high as otherwise the GA will mutate too quickly and have difficulty converging on a solution. It is recommended to set the value between 0.05 and 0.2 for best results.

A comparison of convergence rates for the QBGA with and without the variable mutation rate is shown in Figure A.2. Here it takes only six generations for the varied mutation rate implementation to converge while it takes *twenty-one* generations for the constant rate of mutation implementation to converge to the same fitness value. This indicates that the variable rate of mutation does, indeed, aid in convergence speed as expected.

A.1.2 Fitness Function

The biggest design challenge with GAs in particular is that of the fitness function [45-47, 54]. Care must be taken to add in as many variables as possible in order to eliminate the possibility of unwanted states (for example, a 90° Phase Margin but 10 Hz Bandwidth). However, when more variables are added they will typically need to be weighted in order to emphasize the more important parameters of the circuit.

A way to aide in the convergence of GAs is to add a penalty to the fitness function [47,48]. The idea is to decrease the value of the fitness function based on the region that the variable is in. For example, in order for a circuit to be stable, ϕ_M must be greater than 0°. However, in real-world circuits, having a designed $\phi_M = 0^\circ$ is essentially equivalent to being unstable since components vary and can very easily (and likely *will*) cause the circuit to oscillate. As such, the phase margin should be designed so it is *at minimum* greater than 45°, typically. The lower the phase margin is, the more the circuit 'rings' before settling to its final value which is why 45° is typically chosen as a minimum accepted value. Thus, a penalty function could produce a value for anything below this range to cause the fitness value to decrease more rapidly the further it is away from this point.

A generalized fitness function equation with penalties is shown below in (A.1) where α is a weighted constant for variable θ and $P(\theta)$ is a function that determines the penalty associated with the solution presented by θ .

$$F = \sum_{i=1}^{N} \alpha_i \theta_i - P(\theta_i) \tag{A.1}$$

For this GA implementation, the following variables were selected for the fitness function:

- Phase Margin, ϕ_M
- Gain Margin, G_M
- DC Gain, A_0
- Bandwidth, BW



Figure A.3: Comparison of Convergence Speeds for a) QBGA with NPFF and b) QBGA with Linear Fitness.

Since each of these variables have varying sizes (ϕ_M will be in the sub-100 range while *BW* will get into the 1000s range) it is wise to normalize them to some ideal value such that the sum of all the ideal, normalized values is equal to the number of parameters. This makes it significantly easier to determine the necessary values for α that help to speed convergence.

In addition to normalizing all of the variables, each value was placed into a function that transforms the linear data into parabolic data such that it has a maximum at the ideal point. For example, if the ideal ϕ_M is chosen as 70°, $N_{\phi_M} = \frac{\phi_M}{70}$ and thus $f(\theta(\phi_M)) = -(N_{\phi_M} - 1)^2 + 1$. The constant scalar is there so that the maximum value occurs at a value of 1 instead of 0.

Preliminary results show that the normalized parabolic fitness function, NPFF, exhibits faster global optimum convergence than using only the normalized data alone as shown in Figure A.3. Note that the fitness value in the linear case is, in fact, larger than the NPFF case. This, however, is expected as values *greater* than the normalized ideal parameter will subtract from the fitness value in the NPFF while it adds in the linear case. As such, comparing the fitness values directly does not result in a good comparison of fitness function strength. The only valid comparison is to compare the number of generations it takes to converge.

As Figure A.3 shows, the NPFF takes a mere 4 generations to converge while it takes 8 for the linear function. The simulation was run with 30 drones per generation and a mutation rate of $0.1 \cdot queenAge$ for both simulations. The values for α were



Figure A.4: Step Responses of QBGA with NPFF for a) Generation 1 and b) Generation 10.

 $\alpha_{\phi_M} = 3$, $\alpha_{G_M} = 0.6$, $\alpha_{A_0} = 2$, $\alpha_{BW} = 1.8$ and were determined experimentally. Equation (A.2) shows the full NPFF used for this QBGA implementation.

$$F = \sum_{i=1}^{N} -\alpha_i (N(\theta_i) - 1)^2 + \alpha_i - P(\theta_i)$$
 (A.2)

A.1.3 QBGA Results

The simulation of the QBGA was done on a machine with an Intel Q6600 CPU running at 2.4 GHz with 4 GB of DDR800 RAM at a FSB rate of 1333 MHz. During the simulation, output wave-forms were written to a local 7200 RPM hard disk every five generations. The QBGA had 30 drones per generation and was simulated for 50 generations which had an elapsed time of 130 seconds.

Figure A.4a shows the step response of the first generation. This response is clearly non-ideal as it exhibits ringing for almost a full second. Figure A.4b shows the step response after ten generations where it has fully converged on a solution. This response exhibits great behavior as it has a large phase margin, large gain margin, high DC gain, and adequate bandwidth.

A.2 Particle Swarm Optimization

Another type of optimization algorithm that is useful for high-complexity circuit design is PSO. Using PSO as a circuit design optimization tool has been explored

previously in [60, 61].

This method is different from that of GAs in that there are a set number of particles that have an associated position and velocity in N-dimensional space. Each particle's position is evaluated and each keeps track of its own best position. All of the best positions for each particle are then evaluated and the best global position is stored. Each particle has knowledge of its current position, current velocity, best local position, and best swarm global position. As the swarm begins to move, the velocity of the particles change based on those parameters in order to converge on a solution.

In 2002, Maurice Clerc *et al.* introduced a new way to modify a particle's velocity in order to speed convergence to a solution in [51] and which was later explored more in depth in [56]. This is done by use of constriction factor, χ , and weighted velocity constants, C1 and C2. This constriction factor serves to limit the swarm to a smaller search area as it converges on an optimal solution. The velocity coefficients of C1 and C2 are used to provide more emphasis on a particle's local best solution or the global best solution, respectively. The velocity update equation can be seen in equation (A.3) where β_p is the best local particle position, X_p is the current particle position, and G is the global best position. Equation (A.4) shows the equation to determine the new particle position. Equation (A.5) shows the equation used to calculate the constriction factor, χ .

$$V_p = \chi(V_p + C_1[U(0,1) \cdot (\beta_p - X_p)] + C_2[U(0,1) \cdot (G - X_p)])$$
(A.3)

$$X_p = X_p + V_p \tag{A.4}$$

$$\chi = \left| \frac{2}{2 - (C1 + C2) - \sqrt{(C1 + C2)^2 - 4(C1 + C2)}} \right|$$
(A.5)

The benefit of PSO over GAs is that it is significantly easier to implement as there are no crossover probabilities, mutation probabilities, *etc.* that need to be dealt with. However, the same issues exist with fitness function creation as described in the Genetic Algorithms section which can cause a lot of design issues if not done properly.



Figure A.5: Step Response PSO at a) Iteration 1 and b) Iteration 40.

A.2.1 PSO Results

The fitness function and dimensions simulated are identical in this PSO implementation to the previously described GA implementation and will not be repeated here. The values C1 and C2 were determined, through experimentation, to be 2 and 1.9, respectively. This results in $\chi = 0.95$. Now these values differ from the results found in [56], which provide an ideal C1 + C2 = 4. The reason that C1 + C2 was chosen to be 3.9 was because after thorough investigation, it was discovered that any C1 + C2value above 4 would result in convergence to a local optimum while values less than 4 would slowly converge towards an optimal solution. To fix this issue, the value of 3.9 was chosen as it stays close to the ideal value of 4 while still allowing additional exploration which prevents convergence on a local optimum.

Figure A.5 shows a comparison between the step response in iteration 1 with iteration 40. Here, all parameters are quite good and it's clear that the PSO algorithm has converged on an acceptable solution.

A.3 QBGA and PSO Performance Comparison

Figure A.6 shows a comparison in the convergence rate of both algorithms. As is quite clear, the PSO algorithm takes much longer to converge than the QBGA. Both converge to similar fitness values, but the PSO takes almost $5 \times$ longer to reach the optimal value. This can be attributed in part to the constriction and velocity constant chosen but is more a testament to the robustness of the QBGA with variable mutation rate. As previously shown in Figure A.2, the constant mutation rate converges at



Figure A.6: Fitness Value Convergence Comparisons for QBGA and PSO



Figure A.7: Step Response Comparisons Between a) QBGA and b) PSO for Full System

nearly the same speed as the PSO here which indicates that the variable mutation has a *significant* effect on convergence speed.

As a further comparison, the step response of the full system is shown in Figure A.7. As expected, both the QBGA and PSO converge to similar solutions with relative differences of $\Delta A_0 = 1.9\%$, $\Delta \phi_M = 6.8\%$, $\Delta G_M = 8.2\%$, and $\Delta BW = 0\%$.

It is clear, after thorough experimentation, that utilizing a Queen-Bee Genetic Algorithm with variable mutation rates and a normalized parabolic fitness-function is superior to both a PSO implementation and a QBGA implementation with a constant mutation rate. The quick convergence to an optimal solution indicates that it performs well in a high-dimensional search space. This type of algorithm could benefit from additional research in order to improve functionality and efficiency.

Appendix B

Power Supply Simulation Tutorial with MATLAB[®]

One of the primary issues with switched DC-DC converters is the simulation time involved. There are many options outside of the classical circuit simulation software, all of which do a fine job; however, having the ability to simulate a circuit quickly and accurately within MATLAB[®] is a very powerful tool for a Power Supply Designer. In order to properly simulate a converter, the Simulink and SimScape toolboxes are required. This tutorial will go over the creation of a Controlled Switched DC-DC Boost Converter, but can easily be adapted for any converter topology.

1. To begin, type the command 'powerlib' into the MATLAB[®] command line. This initializes the tool used for circuit creation.

>> powerlib

- 2. An empty window should pop up along with an array of blocks that can be used within the circuit, as shown in Figure B.1. Each block has elements associated with it. For example, the 'Electrical Sources' block has such things as DC voltage sources, Sinusoidal Sources, *etc.*
- 3. Open the 'Elements' box and click on 'Series RLC Branch' and drag it to the circuit window, as shown in Figure B.2.

▶ _a Lit	brary: p	owerlib	_			-					- 0	23
File	Edit	View	Display	Diagram	Analysis	Help						
Þ.,	•				-							
pow	erlib											
۲	Pa pov	verlib 🕨										•
•		+ - Electrica		₩ <u>n</u> ∕_T	Power	Machines	Measurements	Application	Interface	Control	powergui powergui]
-		Sources			Electronics		Sim Pours System	Libraries	Elements	Measurements Library		
»	â					Copyright 1997-2	013 Hydro-Quebec a	and The MathWor	ks, Inc.			-
Read	У						100%					

Figure B.1: Powerlib Toolbar

untitled *	the second se	
File Edit View Display Dia	gram Simulation Analysis Code Tools Help	
B 🗃 💠 🔶 🖥	🖺 🏟 🕶 🝓 🍉 🕪 🗉 💿 👻 10.0 🛛 Normal	• @ • iii •
Model Browser +=	untitled	
Pà untitled	Image: Second	•
	Q	
	•	
	⇒	
	A.	
	▫ᢇᡧᠰ᠆ᡗ᠓ᢆ᠁ᢩᡰᡝ᠆▫	
	Series RLC Branch	
2	«	

Figure B.2: Series RLC Branch

4. Double-click on the Series RLC Branch. Here, the branch type can be changed to R, C, L, or any combination of the three. This is also true for the 'Parallel RLC Branch' within the 'Elements' box.

Select 'L' and enter the desired inductance value. The initial inductor current can also be set in order to speed simulation time, but is not a necessary parameter. This is also true for the output filter capacitor (where it is recommended to set the initial condition to the desired output voltage as this helps speed up the transient simulation tremendously).

5. Next, add in all necessary passive components using the same 'Series RLC Branch' used to generate the inductor. This includes the output filter capacitor, resistive load, and resistive feedback branch. The diode and switch can be found within the 'Power Electronics' box and the voltage supplies and grounds can be found within the 'Electrical Sources' block, as was mentioned earlier. At this point, the circuit should look similar to Figure B.4.

Block Parame	ters: Series RLC Branch
Series RLC Br	anch (mask) (link)
Implements a Use the 'Brand branch.	series branch of RLC elements. h type' parameter to add or remove elements from the
Parameters	
Branch type:	RLC 🔹
Resistance (O	RLC R
1	L
Inductance (H	RL RC
1e-3	LC Open circuit
Set the init	
Capacitance (F):
1e-6	
Set the init	ial capacitor voltage
Measurement	s None 🔹
	OK Cancel Help Apply

Figure B.3: Changing Series RLC Branch Parameters



Figure B.4: Boost Converter in Powerlib (no Controller)

6. The next step is to add in the controller. Unfortunately, a model for the opamp is going to be needed in order to properly simulate the converter. To do so, two voltage measurements are going to be needed (from the 'Measurements' box in the Powerlib tool pane) and an 'Add' block will be needed from the Simulink→Math Operations library. This is the amplifier's differential stage.

- 7. Once the differential stage is implemented, the controller itself must be modeled. The easiest method is to use one of the equations given throughout Chapter 4 to model the controller and implement it with a 'Transfer Fcn' block in the Simulink→Continuous library. At the output of the transfer function, it's wise to play a 'Saturation' block to clamp the amplifier so that it doesn't increase (or decrease) to unrealistic values.
- Finally, add in the 'PWM Generator (DC-DC)' block from the 'Control and Measurements' Library within the Powerlib tool pane. The end circuit should look like the circuit in Figure B.5.



Figure B.5: Boost Converter in Powerlib

- 9. Now that the converter has been entered schematically within Powerlib, measurements must be set up in order to view the output waveforms as shown in Figure B.6. To do so, first place down a 'Voltage Measurement' block near the output of the converter and connected it between the output of the converter and ground. Then place a 'Scope' block from the Simulink→Sinks library and connect it to the output of the Voltage Measurement block. Finally, from the Powerlib tool pane, place a 'Powergui' block anywhere within the circuit. This is required for all Powerlib simulations.
- 10. In order to speed up the simulation as well as guarantee convergence, the simulation solver needs to be changed to **ode23tb**. This is done by navigating to the



Figure B.6: Boost Converter in Powerlib Output Measurement

'Simulation' menu on the menu bar in the Powerlib schematic editor and clicking on 'Model Configuration Parameters'. This brings up a settings window, as shown in Figure B.7. Change the solver to **ode23tb** (stiff/TR-BDF2). The stop time can also be set within this window and was set to 50 ms here.

Select:	Simulation time					
Solver Data Import/Export Data Import/Export Diagnostics Hardware Implementation Model Referencing Simulation Target Code Generation Simscape Simsvechanics 1G SimMechanics 2G	Start time: 0.0		Stop time: 50e-3			
	Solver options Type: Variable-step Max step size: auto Min step size: auto Initial step size: auto Solver reset method: Fast Number of consecutive min steps:		Solver: ode23tb (stiff/TR-BDF2) Relative tolerance: 1e-3 Absolute tolerance: auto Shape preservation: Disable All 1			
	Tasking and sample time options Tasking mode for periodic sample times: Auto Automatically handle rate transition for data transfer Higher priority value indicates higher task priority Zero-crossing options Zero-crossing control: Use local settings Alcorithm: Nonadaptive					
	Time tolerance: Number of consecutive	10*128*eps 2 zero crossings:	Signal threshold:	auto 1000		

Figure B.7: Powerlib Simulation Configuration Settings

11. At this point, the simulation can be run by either clicking the green 'Run' button or by pressing **Ctrl+T**. The output waveform can be viewed by double-clicking

the scope of interest. By navigating to the scope parameters within the scope window, the data can be saved to any variable name in either a Structure or Array format. This is a useful feature that makes post-processing of data very easy.

The simulation for this particular converter took roughly 30 s to simulate 50 ms. This is significantly faster than conventional circuit simulation software which can take $10 \times$ as long, if not longer, which is what makes supply simulation in MATLAB[®] a very attractive option for designers.

References

- [1] Yang-Ching Lin, Yean-Kuo Luo, Ke-Horng, Chen, Wei-Chou Hsu, "Liquid Crystal Display Supplied by Highly Integrated Dual-Side Dual-Output Switched Capacitor DC-DC Converter With Only Two Flying Capacitors," in *IEEE Transactions on Circuits and Systems - I: Regular Papers*, vol. 59, no. 2, pp. 439-446, February 2012.
- [2] Feng Luo, Dongsheng Ma, "An Integrated Switching DC-DC Converter With Dual-Mode Pulse-Train/PWM Control," in *IEEE Transactions on Circuits and Systems - II: Express Briefs*," vol.56, no. 2, pp. 152-156, February 2009.
- [3] Jeongjin Roh, "High-Performance Error Amplifier for Fast Transient DC-DC Converters," in *IEEE Transactions on Circuits and Systems-II: Express Briefs*, vol. 52, no. 9, pp. 591-595, September 2005.
- [4] Hong-Wei Huang, Hsin-Hsin Ho, Chia-Jung Chang, Ke-Horng Chen, and Sy-Yen Kuo, "One-Chip Compensated Error Amplifier for Fast Transient DC-DC Converters," in *IEEE International Conference on Electro/information Technology*, East Lansing, MI, pp. 103-108. May 7-10, 2006.
- [5] Tsz Yin Man, Philip K.T. Mok, Mansun J. Chan, "A 0.9-V Input Discontinuous-Conduction-Mode Boost Converter with CMOS-Control Rectifier," in *IEEE Journal of Solid-State Circuits*, vol. 43, no. 9, pp. 2036-2046, September 2008.
- [6] Chan-Soo Lee, Young-Jin Oh, Kee-Yeol Na, Yeong-Seuk Kim, Nam-Soo Kim "Integrated BiC-MOS Control Circuits for High-Performance DC-DC Boost Converter," in *IEEE Transactions on Power Electronics*, vol. 28, no. 5, pp. 2596-2603, May 2013.
- [7] Se-Won Wang, Young-Jin Woo, Sung-Ho Bae, Tae-Hwang Kong, Gyu-Ha Cho, Cyu-Hyeong Cho, "A High Stability DC-DC Boost Converter with Ripple Current Control and Capacitor-Free LDOs for AMOLED Display," in *IEEE Asian Solid-State Circuits Conference* pp. 41-44, November 2011.
- [8] Mauro Olivieri, Roberto Mancuso, Friedbert Riedel, "A Reconfigurable, Low Power, Temperature Compensated IC for 8-segment Gamma Correction Curve in TFT, OLED, and PDP

Displays," in *IEEE Transactions on Consumer Electronics* vol. 53, no. 2, pp. 720-724, May 2007.

- [9] Min-Hau Michael Lu, Michael Hack, Richard Hewitt, Michael S. Weaver, Julia J. Brown, "Power Consumption and Temperature Increase in Large Area Active-Matrix OLED Displays," in *Journal of Display Technology*, vol. 4, no. 1, pp. 47-53, March 2008.
- [10] Dariusz Czarkowski, and M.K. Kazimierczuk, "Energy-Conservation Approach to Modeling PWM DC-DC Converters," in *IEEE Transactions on Aerospace and Electronic Systems*, vol. 29, no. 3, pp. 1059-1063, 1993.
- Travis Eichhorn, "Boost Converter Efficiency Through Accurate Calculations," in *Power Electronics Technology*, pp. 30-35, September 2008.
- [12] Tony Chan Carusone, David A. Johns, Kenneth W. Martin, "Integrated-Circuit Devices and Modeling," in Analog Integrated Circuit Design, 2nd ed. Hoboken, NJ: Wiley, ch. 1, pp.10-13; 62-64, 2012.
- [13] Daniel W. Hart, *Power Electronics*, 1st ed. New York, NY: McGraw-Hill, 2011.
- [14] R.D. Middlebrook, and Slobodan Ćuk, "A General Unified Approach to Modeling Switching-Converter Power Stages," in *Proceedings of the IEEE Power Electronics Specialists Conference*, pp. 18-34, 1976.
- [15] Slobodan Čuk, "Modeling, Analysis, and Design of Switching Converters," Ph.D. Dissertation, Department of Engineering and Applied Science, California Institute of Technology, Pasadena, CA, November 29, 1976.
- [16] R.D. Middlebrook, "Small-Signal Modeling of Pulse-Width Modulated Switched-Mode Power Converters," in *Proceedings of the IEEE*, vol. 76, no. 4, pp. 343-354, April 1988.
- [17] Vandana Jha, and Pankaj Rai, "State Space Averaged Modeling of Basic Converter Topologies," in VSRD International Journal of Electrical, Electronics, and Communication Engineering, vol. 2, no. 8, pp. 566-575, 2012.
- [18] Vatché Vorpérian, "Simplified Analysis of PWM Converters Using Model of PWM Switch Part I: Continuous Conduction Mode," in *IEEE Transactions on Aerospace and Electronic Systems*, vol. 26, no. 3, pp. 490-496, May 1990.
- [19] Vatché Vorpérian, "Simplified Analysis of PWM Converters Using Model of PWM Switch Part II: Discontinuous Conduction Mode," in *IEEE Transactions on Aerospace and Electronic Systems*, vol. 26, no. 3, pp. 497-505, May 1990.
- [20] Alberto Reatti, and M.K. Kazimierczuk, "Current-Controlled Current Source Model for a PWM DC-DC Boost Converter Operated in Discontinuous Conduction Mode," in *IEEE International Symposium on Circuits and Systems*, Geneva, Switzerland, May 28-31, 2000, pp. III/239-III/242.
- [21] Christophe P. Basso, Switched-Mode Power Supplies, 1st ed. New York, NY: McGraw-Hill, 2008.
- [22] Yannis Tsividis, and Colin McAndrew, "The Four-Terminal MOS Transistor," in Operation and Modeling of the MOS Transistor, 3rd ed. New York, NY: Oxford, ch. 4, pp. 186-210, 2011.
- [23] Robert W. Erickson, and Dragan Maksimović, Fundamentals of Power Electronics, 2nd ed. Secaucus, NJ: Academic, 2000.
- [24] Robert Mammano, "Switching Power Supply Topology: Voltage Mode vs. Current Mode," in Unitrode Application Notes, DN-62, Texas Instruments, 1999.
- [25] R. D. Middlebrook, "Topics in Multiple-Loop Regulators and Current-Mode Programming," in *IEEE Transactions on Power Electronics*, vol. 2, no. 2, pp. 109-124, April 1987.
- [26] Raymond B. Ridley, "A New Continuous-Time Model for Current-Mode Control," in *IEEE Transactions on Power Electronics*, vol. 6, no. 2, pp. 271-280, April 1991.
- [27] Yushan Li, Kevin R. Vannorsdel, Art J. Zirger, Mark Norris, and Dragan Maksimović, "Current Mode Control for Boost Converters with Constant Power Loads," in *IEEE Transactions on Circuits and Systems-I: Regular Papers*, vol. 59, no. 1, pp. 198-206, January 2012.
- [28] Cheung Fai Lee and Philip K.T. Mok, "A Monolithic Current-Mode CMOS DC-DC Converter With On-Chip Current-Sensing Technique," in *IEEE Journal of Solid-State Circuits*, vol. 39, no. 1, pp. 3-14, January 2004.
- [29] Hong Yao, "Modeling and Design of a Current Mode Control Boost Converter," M.S. Thesis, Department of Electrical and Computer Engineering, Colorado State University, Fort Collins, CO, 2012.
- [30] Michael Qiao, Parvitz Parto, and Reza Amirani, "Stabilizing Buck Converters with Transconductance Amplifiers," in *EE Times*, Internet: http://www.eetimes.com/electronicsnews/4164406/Stabilizing-buck-converters-with-transconductance-amplifiers, November 13, 2002 [June 15, 2013].
- [31] Christophe P. Basso, "Transient Response Counts When Choosing Phase Margin," Power Electronics Technology, pp. 18-21, November 2008.

- [32] R.D. Middlebrook, "Measurement of Loop Gain in Feedback Systems," in International Journal of Electronics, vol. 38, no. 4, pp. 485-512. September 1975.
- [33] Ray Ridley, "Transient Response and Loop Gains of Power Supplies," in Switching Power Magazine, pp. 2-6, 2005.
- [34] Botao Miao, Regan Zane, Dragan Maksimović, "System Identification of Power Converters with Digital Control Through Cross-Correlation Methods," in *IEEE Transactions on Power Electronics*, vol. 20, no. 5, pp. 1093-1099, September 2005.
- [35] C. Fernández, A. Fernández-Herrero, P. Zumel, A. Lázaro, A. Barrado, "Measuring Bode Plots of Switching Power Converters from a Single Simulation in the Time Domain: Application to a Digital Control Implemented on an FPGA," in *IEEE 12th Workshop on Control and Modeling* for Power Electronics, Boulder, CO, pp. 1-7, June 2010.
- [36] Adam Barkley, Enrico Santi, "Improved Online Identification of a DC-DC Converter and Its Control Loop Gain Using Cross-Correlation Methods," in *IEEE Transactions on Power Electronics*, vol. 24, no. 8, pp. 2021-2031, August 2009.
- [37] Mariko Shirazi, Jeffrey Morroni, Arseny Dolgov, Regan Zane, Dragan Maksimović, "Integration of Frequency Response Measurement Capabilities in Digital Controllers for DC-DC Converters," in *IEEE Transactions on Power Electronics*, vol. 23, no. 5, pp. 2524-2535, September 2008.
- [38] Fran González-Espín, Emilio Figueres, Gabriel Garcerá, Raúl González-Medina, Marcos Pascual, "Measurement of the Loop Gain Frequency Response of Digitally Controlled Power Converters," in *IEEE Transactions on Industrial Electronics*, vol. 57, no. 8, pp. 2785-2795, August 2010.
- [39] Marcel Allain, Philippe Viarouge, Faouzi Tourkhami, "The use of Pseudo-Random Binary Sequences to Predict a DC-DC Converter's Control-to-Output Transfer Function in Continuous Conduction Mode," in *Canadian Conference on Electrical and Computer Engineering*, pp. 574-577, Saskatoon, Sask., May 2005.
- [40] Mor Mordechai, Sam Ben-Yaakov, "Time Domain Identification of PWM Converters for Digital Controllers Design," in *IEEE Power Electronics Specialists Conference*, pp. 809-813, Orlando, FL, June 2007.
- [41] Bengt Johansson, Matz Lenells, "Possibilities of Obtaining Small-Signal Models of DC-to-DC Power Converters by Means of System Identification," in 22nd International Telecommunications Energy Conference, pp. 65-75, Phoenix, AZ, 2000.
- [42] F. Jessie MacWilliams, Neil J.A. Sloane, "Pseudo-Random Sequences and Arrays," in *Proceed-ings of the IEEE*, vol. 64, no. 12, pp. 1715-1729, December 1976.

- [43] Lance C. Benn, Bruce Burton, Ron G. Harley, "PWM Spectrum Analysis and Optimization for Model Based Condition Monitoring and Control," in *Industry Applications Conference*, vol. 2, pp. 1311-1318, Chicago, IL, October 2001.
- [44] Dean Venable, "Testing Power Sources for Stability," Unpublished.
- [45] Günter Rudolph, "Convergence Analysis of Canonical Genetic Algorithms," in *IEEE Transac*tion on Neural Networks, vol. 5, no. 1, pp. 96-101, January 1994.
- [46] Jeffrey A. Joines, Christopher R. Houck, "On the Use of Non-Stationary Penalty Functions to Solve Nonlinear Constrained Optimization Problems with GA's," in *Proceedings of the First IEEE Conference on Evolutionary Computation* vol. 2, pp. 579-584, June 1994.
- [47] Mitsuo Gen, Runwei Cheng, "A Survey of Penalty Techniques in Genetic Algorithms," in Proceedings of IEEE International Conference on Evolutionary Computation, pp. 804-809, May 1996.
- [48] Vassilios Petridis, Spyros Kazarlis, Anastasios Bakirtzis, "Varying Fitness Functions in Genetic Algorithm Constrained Optimization: The Cutting Stock and Unit Commitment Problems," in *IEEE Transactions on Systems, Man, and Cybernetics - Part B: Cybernetics*, vol. 28, no. 5, pp. 629-640, October 1998.
- [49] Adrian Thompson, Paul Layzell, Ricardo Salem Zebulum, "Explorations in Design Space: Unconventional Electronics Design Through Artificial Evolution," in *IEEE Transactions on Evolutionary Computation*, vol. 3, no. 3, pp. 167-196, September 1999.
- [50] Jun Zhang, Henry Shu-Hung Chung, Wai-Lun Lo, S.Y. Hui, Angus Kwok-Ming Wu, "Implementation of a Decoupled Optimization Technique for Design of Switching Regulators Using Genetic Algorithms," in *IEEE Transactions on Power Electronics*, vol. 16, no. 6, pp. 752-763, November 2001.
- [51] Maurice Clerc, James Kennedy, "The Particle Swarm Explosion, Stability, and Convergence in a Multidimensional Complex Space," in *IEEE Transactions on Evolutionary Computation*, vol. 6, no. 1 pp. 58-73, February 2002.
- [52] Sung Hoon Jung, "Queen-bee Evolution for Genetic Algorithms," in *Electron Letters* vol. 39, no. 6, pp. 575-576, March 2003.
- [53] Kazuhiro Takemura, Tetsushi Koide, Hans Jürgen Mattausch, Toshio Tsuji, "Analog-Circuit-Component Optimization with Genetic Algorithm," in *IEEE International Midwest Sympo*sium on Circuits and Systems, vol. 1, pp. 489-492, July 2004.

- [54] Shinn-Ying Ho, Li-Sun Shu, Jian-Hung Chen, "Intelligent Evolutionary Algorithms for Large Parameter Optimization Problems," in *IEEE Transactions on Evolutionary Computation*, vol. 8, no. 6, pp. 522-541, December 2004.
- [55] Jun Zhang, Henry S.H. Chung, W.L. Lo, "Pseudocoevolutionary Genetic Algorithms for Power Electronics Circuits Optimization," in *IEEE Transactions on Systems, Man, and Cybernetics*, vol. 36, no. 4, pp. 590-598, July 2006.
- [56] Daniel Bratton, James Kennedy, "Defining a Standard for Particle Swarm Optimization," in Proceedings of the IEEE Swarm Intellegence Symposium, pp. 120-127, April 2007.
- [57] Zhiguo Bao, Takahiro Watanabe, "A New Approach for Circuit Design Optimization Using Genetic Algorithm," in *International SoC Design Conference*, vol. 1, pp. I383-I386, November 2008.
- [58] Zhiguo Bao, Takahiro Watanabe, "A Novel Genetic Algorithm with Cell Crossover for Circuit Design Optimization," in *IEEE International Symposium on Circuits and Systems*, pp. 2982-2985, May 2009.
- [59] Kinattingal Sundareswaren, V.T. Sreedevi, "Boost Converter Controller Design Using Queen-Bee-Assisted GA," in *IEEE Transactions on Industrial Electronics*, vol. 56, no. 3, pp. 778-783, March 2009.
- [60] Ahmed F. Zobaa, Agostino Lecci, "Power Circuit Design Based on PSO Optimization," in International Power Electronics and Motion Control Conference, pp. 121-125, September 2010.
- [61] M. Kotti, B. Benhala, M. Fakhfakh, A. Ahaitouf, B. Benlahbib, M. Loulou, A. Mecheqrane, "Comparison between PSO and ACO Techniques for Analog Circuit Performance Optimization," in *International Conference on Microelectronics*, pp. 1-6, December 2011.
- [62] Xuejan Song, Yanli Cui, Aiting Li, "Optimization Algorithm of Evolutionary Design of Circuits Based on Genetic Algorithm," in *Fifth International Symposium on Computational Intelligence* and Design, vol. 1, pp. 336-339, October 2012.